REPRINT

COMPANIES (INSOLVENCY AND RECEIVERSHIP) ACT 2009

As in force at: 5 February 2024

STATUS: CURRENT

For details see Endnotes

AN ACT TO PROVIDE FOR THE MANAGEMENT AND ADMINISTRATION OF INSOLVENT COMPANIES AND TO REFORM THE LAW RELATING TO RECEIVERS IN SOLOMON ISLANDS

COMPANIES (INSOLVENCY AND RECEIVERSHIP) ACT 2009

Table of provisions

PART 1		PRELIMINARY PROVISIONS	1
1 2 3 4 5	Interpre Purpose	encement	1 1 1 1
PART 2		INSOLVENT COMPANIES	2
Division	1	Compromises with Creditors	2
6 7 8 9 10 11	Compromise proposal Notice of proposed compromise Effect of compromise Variation of compromise Powers of Court Effect of compromise in liquidation of company Costs of compromise		
Division	2	Liquidations	6
Subdivis	sion A	Purpose	6
13	Purpose	e	6
Subdivis	ion B	Beginning of Liquidation	7
14	When li	quidation begins	7
Subdivis	ion C	Restrictions on Appointment of Liquidator	7
15	Restrict	ions on appointment of liquidator	7
Subdivis	ion D	How Liquidator may be Appointed	8
16 17		nay appoint liquidator olders may appoint liquidator	8

18 19 20 21 22 23 24	Court may appoint liquidator Interim liquidator Meaning of unable to pay its debts Evidence and other matters Statutory demand Court may set aside statutory demand Additional powers of Court on application to set aside statutory demand		
Subdivi	sion E Notices	13	
25 26	Notices given by liquidator Documents to state company in liquidation	13 13	
Subdivi	sion F Obligations to Liquidators	13	
27 28	Directors, etc. to identify and deliver company property Obligations of suppliers of essential services	13 14	
Subdivi	sion G Liquidators' Rights to Company's Documents	14	
29 30 31 32 33 34 35 36 37	Liquidator may require director, etc. to deliver documents Liquidator may require director, etc. to provide information Reasonable expenses may be paid Examination by liquidator Court may order person to comply with section 30 Self-incrimination no excuse Restriction on enforcement of lien over company's documents Delivery of document held by secured creditor Documents held by receiver	14 14 15 16 16 17 17 17	
Subdivi	sion H Meetings	18	
38 39 40 41 42 43 44 45	Notice of first creditors' meeting Timing of first creditors' meeting Purpose of first creditors' meeting Replacement liquidator Effect of directors' resolving company able to pay its debts Other creditors' meetings Liquidator may dispense with meetings of creditors Meetings of creditors or shareholders Views of creditors and shareholders at meetings to be considered	18 19 19 20 20 21 22	

Subdivis	ion I	Reports	24
47 48 49	First report Six-monthly report Exemption from reporting requirements		
Subdivis	ion J	Creditors' Claims	25
50 51		ntial claims of other creditors and distribution of surplus assets	25 26
Subdivis	ion K	End of Liquidation	27
52 53 54 55 56 57 58	Completion of liquidation Final report and accounts Liquidation surplus account Termination of liquidation by Court Notice of termination of liquidation Liquidation of assets of overseas company in Solomon Islands Liquidation of assets of community companies		
PART 3		RECEIVERSHIPS	31
Division	1	Application	31
59 60		ion of this Part to receivers ion of this Part to receivers	31 31
Division	2	Appointment of Receivers	32
61 62 63	Appoint	ions on appointment of receiver ment of receiver under document of power to appoint receiver under document	32 32 32
Division	3	Notices	33
64 65		given by receiver of receivership	33 33
Division	4	Obligations to Receivers	34
66 67	_	ons of grantor ons of suppliers of essential services	34 34

Division	5 Reports of Receivers	35
68 69 70 71 72 73 74	First report of receiver Contents of first report Further reports of receiver Contents of further reports What may be omitted from reports Extension of time for preparing reports Persons entitled to receive reports Persons entitled to inspect reports	35 36 37 38 38 38
Division	6 Preferential Claims	39
76 77	Preferential claims Receiver appointed under floating security interest: ranking of preferential claims	39 40
Division	7 End of Receivership	40
78 79	Notice of end of receivership Court may terminate or limit receivership	40 41
PART 4	MISCELLANEOUS	42
81	Regulations	42
SCHED	JLE 1 INTERPRETATION	44
SCHED	JLE 2 MEETINGS OF CREDITORS	52
SCHED	JLE 3 POWERS, FUNCTIONS, AND LIABILITIES OF LIQUIDATORS	61
SCHEDI	JLE 4 OFFICE OF LIQUIDATOR	72 72
SCHEDI	JLE 5 EFFECT OF LIQUIDATION	76
SCHED	JLE 6 LIQUIDATION COMMITTEES	81

SCHEDULE 7	85
VOIDABLE TRANSACTIONS AND SECURITY INTERESTS AND RECOVERIES IN OTHER CASES	85
SCHEDULE 8	103
CREDITORS' CLAIMS	103
SCHEDULE 9	
LIQUIDATION OF ASSETS OF OVERSEAS COMPANIES	119
SCHEDULE 10	
POWERS, DUTIES, AND LIABILITIES OF RECEIVERS	120
SCHEDULE 11	
OFFICE OF RECEIVER	132

COMPANIES (INSOLVENCY AND RECEIVERSHIP) ACT 2009

As in force at: 5 February 2024

PART 1 PRELIMINARY PROVISIONS

1 Short title

This Act may be cited as the Companies (Insolvency and Receivership) Act 2009.

2 Commencement

This Act commences on the same date of commencement of the *Companies Act 2009*.

3 Interpretation

Definitions and other interpretation provisions that apply to this Act are set out in Schedule 1.

4 Purpose of each Part

The purposes of each Part are:

- (a) Part 1 deals with the name of this Act, the reference to definitions set out in Schedule 1 and the application of this Act to the Crown; and
- (b) Part 2 contains various procedures for insolvent companies, including compromises with creditors and liquidations; and
- (c) Part 3 deals with receiverships and the powers of receivers; and
- (d) Part 4 sets out the miscellaneous provision on regulationmaking powers.

5 Act binds the Crown

This Act binds the Crown.

PART 2 INSOLVENT COMPANIES

Division 1 Compromises with Creditors

6 Compromise proposal

Any of the following persons may propose a compromise under this Division if that person has reason to believe that a company is or will be unable to pay its debts as they become due in the normal course of business:

- (a) the directors of the company;
- (b) a receiver appointed in relation to the whole or substantially the whole of the assets and undertaking of the company;
- (c) a liquidator of the company.

7 Notice of proposed compromise

- (1) The proponent must compile, in relation to each class of creditors of the company, a list of creditors known to the proponent who would be affected by the proposed compromise, setting out:
 - (a) the amount owing or estimated to be owing to each of them; and
 - (b) the number of votes that each of them is entitled to cast on a resolution to approve the compromise.
- (2) The proponent must give to each known creditor, the company, any receiver or liquidator, and deliver to the Registrar for registration:
 - (a) notice in accordance with Schedule 2 of the intention to hold a meeting of creditors, or any 2 or more classes of creditors, for the purpose of voting on the resolution; and
 - (b) a statement:
 - (i) containing the name and address of the proponent and the capacity in which the proponent is acting; and
 - (ii) containing the address and telephone number to which

- inquiries may be directed during normal business hours; and
- (iii) setting out the terms of the proposed compromise and the reasons for it; and specifying (where applicable) the property of the company that is available to pay creditors' claims; the duration of any moratorium period; the extent to the which the company is released from its debts; the conditions (if any) of the compromise to commence, continue or terminate; the order of distribution of proceeds amongst creditors; the cut-off date for claims to be included; and
- (iv) setting out the reasonably foreseeable consequences for creditors of the company of the compromise being approved; and
- (v) setting out the extent of any interest of a director in the proposed compromise; and
- (vi) explaining that the proposed compromise and any amendment to it proposed at a meeting of creditors or any classes of creditors will be binding on all creditors, or on all creditors of that class, if approved in accordance with section 8; and
- (vii) containing details of any procedure proposed as part of the proposed compromise for varying the compromise following its approval; and
- (c) a copy of the list or lists of creditors referred to in subsection (1).

8 Effect of compromise

- (1) A compromise, including any amendment proposed at the meeting, is approved by creditors, or a class of creditors, if, at a meeting of creditors or that class of creditors conducted in accordance with Schedule 2, the compromise, including any amendment, is adopted in accordance with that schedule.
- (2) A compromise, including any amendment, approved by creditors or

- a class of creditors of a company in accordance with this Division is binding on the company and on all creditors, or, if there is more than 1 class of creditors, on all creditors of that class, to whom notice of the proposal was given.
- (3) If a resolution proposing a compromise, including any amendment, is put to the vote of more than 1 class of creditors, it is to be presumed, unless the contrary is expressly stated in the resolution, that the approval of the compromise, including any amendment, by each class is conditional on the approval of the compromise, including any amendment, by every other class voting on the resolution.
- (4) The proponent must give written notice of the result of the voting to each known creditor, the company, any receiver or liquidator, and the Registrar.

9 Variation of compromise

- (1) An approved compromise may be varied or terminated either:
 - (a) in accordance with any procedure for variation or termination incorporated in the compromise as approved; or
 - (b) by the approval of a proposal to vary or terminate the compromise in accordance with this Division that, for that purpose, applies, with all necessary modifications, as if any the proposal were a proposed compromise.
- (2) This Division applies to any compromise that is varied in accordance with this section.

10 Powers of Court

- (1) On the application of the proponent or the company, the Court may:
 - (a) give directions in relation to a procedural requirement imposed by this Division, or waive or vary any such requirement, if it is satisfied that it would be just to do so; or
 - (b) order that, during a period specified in the order, beginning not earlier than the date on which notice was given of the proposed

compromise and ending not later than 10 working days after the date on which notice was given of the result of the voting on it:

- (i) proceedings in relation to a debt owing by the company be stayed; or
- (ii) a creditor refrain from taking any other measure to enforce payment of a debt owing by the company.
- (2) Nothing in subsection (1)(b) affects the right of a secured creditor during that period to take possession of, realise, or otherwise deal with, property of the company over which that creditor has a security interest.
- (3) The Court may order that the creditor is not bound by the compromise or make any other order that it thinks fit if the Court is satisfied, on the application of a creditor of a company who was entitled to vote on a compromise, that:
 - (a) not enough notice of the meeting or of the matter required to be notified under section 7 was given to that creditor; or
 - (b) there was some other material irregularity in obtaining approval of the compromise; or
 - (c) in the case of a creditor who voted against the compromise, the compromise is unfairly prejudicial to that creditor, or to the class of creditors to which that creditor belongs.
- (4) An application under subsection (3) must be made not later than 10 working days after the date on which notice of the result of the voting was given to the creditor.

11 Effect of compromise in liquidation of company

- (1) If a compromise is approved, the Court may, on the application of:
 - (a) the company; or
 - (b) a receiver appointed in relation to property of the company; or
 - (c) with the leave of the Court, any creditor or shareholder of the company,

make such order as the Court thinks fit with respect to the extent, if any, to which the compromise will, if the company is put into liquidation, continue in effect and be binding on the liquidator of the company.

(2) If a compromise is approved and the company is later put into liquidation, the Court may, on the application of the liquidator or any person described in subsection (1)(b) or (c), make any order that the Court thinks fit with respect to the extent, if any, to which the compromise will continue in effect and be binding on the liquidator of the company.

12 Costs of compromise

Unless the Court orders otherwise, the costs incurred in organising and conducting a meeting of creditors for the purpose of voting on a proposed compromise:

- (a) must be met by the company; or
- (b) if incurred by a receiver or a liquidator, are a cost of the receivership or liquidation; or
- (c) if incurred by any other person, are a debt due to that person by the company.

Division 2 Liquidations

Subdivision A Purpose

13 Purpose

- (1) The purpose of this Division and Schedules 2 to 8 is to provide for a liquidator of a company:
 - to take possession of, protect, realise, and distribute the assets, or the proceeds of the realisation of the assets, of the company to its creditors in accordance with this Act; and
 - (b) if there are surplus assets remaining, to distribute them, or the proceeds of the realisation of the surplus assets, in accordance with section 51.

- (2) The provisions relating to liquidations are set out in:
 - (a) this Division; and
 - (b) Schedule 2, which, with the necessary modifications, applies to meetings of creditors; and
 - (c) Schedules 3 and 4, which apply to liquidators; and
 - (d) Schedule 5, which sets out the effect of liquidation; and
 - (e) Schedule 6, which applies to liquidation committees; and
 - (f) Schedule 7, which applies to voidable transactions and security interests and recoveries in other cases in a liquidation; and
 - (g) Schedule 8, which applies to creditors' claims.

Subdivision B Beginning of Liquidation

14 When liquidation begins

- (1) The liquidation of a company begins on the date on which, and at the time at which, the liquidator is appointed.
- (2) The liquidator must be a named person.
- (3) If any question arises as to whether, on the date on which a liquidator was appointed, an act was done or a transaction was entered into or effected before or after the time at which the liquidator was appointed, that act or transaction is, in the absence of proof to the contrary, deemed to have been done or entered into or effected, as the case may be, after that time.

Subdivision C Restrictions on Appointment of Liquidator

15 Restrictions on appointment of liquidator

- (1) Unless the Court orders otherwise, none of the persons referred to in clause 1 of Schedule 4 may be appointed or act as a liquidator of a company.
- (2) The appointment of a person as a liquidator is of no effect unless that

person has consented in writing to the appointment.

(3) A person who acts as a liquidator in contravention of subsection (1) commits an offence and is liable on conviction to a fine not exceeding 2,500 penalty units.

Subdivision D How Liquidator may be Appointed

16 Board may appoint liquidator

- (1) A liquidator may be appointed by the resolution of the board of directors of the company on the occurrence of an event specified in the company's rules.
- (2) The board of directors of the company must record, in the document appointing the liquidator, the date on which, and the time at which, the liquidator was appointed.

17 Shareholders may appoint liquidator

- (1) A liquidator may be appointed by special resolution of those shareholders entitled to vote and voting on the question.
- (2) The shareholders must record in the special resolution appointing the liquidator the date on which, and the time at which, the special resolution was passed.

18 Court may appoint liquidator

- (1) A liquidator may be appointed by the Court on the application of:
 - (a) the company; or
 - (b) a director of the company; or
 - (c) a shareholder of the company; or
 - (d) a creditor of the company (including any contingent or prospective creditor); or
 - (e) the Registrar.
- (2) The Court may appoint a liquidator if it is satisfied that:

- (a) the company is unable to pay its debts; or
- (b) the company or the directors have persistently or seriously failed to comply with this Act; or
- (c) it is just and equitable that the company be put into liquidation.
- (3) The Court must record in the order appointing the liquidator the date on which, and the time at which, the order was made.

19 Interim liquidator

- (1) If an application has been made to the Court for an order that a company be put into liquidation, the Court may, if it is satisfied that it is necessary or expedient for the purpose of maintaining the value of assets owned or managed by the company, appoint a named person as interim liquidator.
- (2) Subject to subsection (3), an interim liquidator has the rights and powers of a liquidator to the extent necessary or desirable to maintain the value of assets owned or managed by the company.
- (3) The Court may limit the rights and powers of an interim liquidator in the manner that it thinks fit.
- (4) The appointment of an interim liquidator takes effect on the date on which, and at the time at which, the order appointing that interim liquidator is made.
- (5) The Court must record in the order appointing the interim liquidator the date on which, and the time at which, the order was made.
- (6) If any question arises as to whether, on the date on which an interim liquidator was appointed, an act was done or a transaction was entered into or effected before or after the time at which the interim liquidator was appointed, that act or transaction is, in the absence of proof to the contrary, deemed to have been done or entered into or effected, as the case may be, after that time.

20 Meaning of unable to pay its debts

Unless the contrary is proved, and subject to section 21, a company is presumed to be unable to pay its debts if:

- (a) the company has failed to comply with a statutory demand; or
- (b) execution issued against the company in respect of a judgment debt has been returned unsatisfied in whole or in part; or
- (c) a person entitled to a security interest over all or substantially all of the property of the company has appointed a receiver under the document creating the security interest; or
- (d) a compromise between a company and its creditors has been put to a vote in accordance with this Division.

21 Evidence and other matters

- (1) On an application to the Court for an order that a company be put into liquidation, evidence of failure to comply with a statutory demand is not admissible as evidence that a company is unable to pay its debts unless the application is made within 30 working days after the last date for compliance with the demand.
- (2) Section 20 does not prevent proof by other means that a company is unable to pay its debts.
- (3) Information or records acquired under section 57 of the *Companies Act 2009* or, if the Court so orders, under section 62 of the *Companies Act 2009*, may be received as evidence that a company is unable to pay its debts.
- (4) In determining whether a company is unable to pay its debts, contingent or prospective liabilities may be taken into account.
- (5) An application to the Court for an order that a company be put into liquidation on the ground that it is unable to pay its debts may be made by a contingent or prospective creditor only with the leave of the Court; and the Court may give such leave, with or without conditions, only if it is satisfied that a *prima facie* case has been made out that the company is unable to pay its debts.

22 Statutory demand

(1) A statutory demand is a demand by a creditor in respect of a debt owing by a company made in accordance with this section.

- (2) A statutory demand must:
 - (a) be in respect of a debt that is due and payable and is not less than the prescribed amount; and
 - (b) be in writing; and
 - (c) be served on the company; and
 - (d) require the company to do any of the following things to the reasonable satisfaction of the creditor, within 15 working days of the date of service, or any longer period that the Court may order:
 - (i) pay the debt;
 - (ii) enter into a compromise under Division 1;
 - (iii) otherwise compound with the creditor;
 - (iv) give a security interest over its property to secure payment of the debt.

23 Court may set aside statutory demand

- (1) The Court may, on the application of the company, set aside a statutory demand.
- (2) The application must be:
 - (a) made within 10 working days of the date of service of the demand; and
 - (b) served on the creditor within 10 working days of the date of service of the demand.
- (3) No extension of time may be given for making or serving an application to have a statutory demand set aside, but, at the hearing of the application, the Court may extend the time for compliance with the statutory demand.
- (4) The Court may grant an application to set aside a statutory demand if it is satisfied that:

- (a) there is a substantial dispute as to whether or not the debt is owing or is due; or
- (b) the company appears to have a counterclaim, set-off, or crossdemand and the amount specified in the demand less the amount of the counterclaim, set-off, or cross-demand is less than the prescribed amount; or
- (c) the demand ought to be set aside on other grounds.
- (5) A demand must not be set aside by reason only of a defect or irregularity unless the Court considers that substantial injustice would be caused if it were not set aside.
- (6) In subsection (5), "defect" includes a material misstatement of the amount due to the creditor and a material misdescription of the debt referred to in the demand.
- (7) An order under this section may be made subject to conditions.

24 Additional powers of Court on application to set aside statutory demand

- (1) If, on the hearing of an application under section 23, the Court is satisfied that there is a debt due by the company to the creditor that is not the subject of a substantial dispute, or is not subject to a counterclaim, set-off, or cross-demand, the Court may, on the ground that the company is unable to pay its debts:
 - (a) order the company to pay the debt within a specified period and that, in default of payment, the creditor may make an application to put the company into liquidation; or
 - (b) dismiss the application and immediately make an order putting the company into liquidation.
- (2) For the purposes of the hearing of an application to put the company into liquidation under an order made under subsection (1)(a), the company is presumed to be unable to pay its debts if it failed to pay the debt within the specified period under the order.

Subdivision E Notices

25 Notices given by liquidator

- (1) A liquidator must, immediately after being appointed or being notified of his or her appointment, give public notice of:
 - (a) the liquidator's appointment; and
 - (b) the date and time of the commencement of the liquidation; and
 - (c) the address, telephone and fax numbers, and email address to which, during normal business hours, inquiries may be directed by a creditor or shareholder.
- (2) A liquidator must, within 10 working days of being appointed or being notified of his or her appointment, deliver to the Registrar for registration a notice of the liquidator's appointment.

26 Documents to state company in liquidation

Any document and negotiable instrument entered into, made, or issued by a liquidator of a company on behalf of the company must state, in a prominent position, that the company is in liquidation.

Subdivision F Obligations to Liquidators

27 Directors, etc. to identify and deliver company property

- (1) A present or former director or employee of a company in liquidation must:
 - immediately after the company is put into liquidation, give the liquidator details of property of the company in his or her possession or under his or her control; and
 - (b) on being required to do so by the liquidator, immediately or within any time that may be specified by the liquidator, deliver the property to the liquidator or any other person that the liquidator may direct, or dispose of the property in any manner that the liquidator may direct.
- (2) A person who fails to comply with subsection (1) commits an offence

and is liable on conviction to a fine not exceeding 12,500 penalty units or to imprisonment for a term not exceeding 2 years, or both.

28 Obligations of suppliers of essential services

- (1) Despite any other written law or any contract, a supplier of an essential service must not:
 - (a) refuse to supply the service to a liquidator, or to a company in liquidation, by reason of the company's default in paying charges due for the service in relation to a period before the commencement of the liquidation; or
 - (b) make it a condition of the supply of the service to a liquidator, or to a company in liquidation, that payment be made of outstanding charges due for the service in relation to a period before the commencement of the liquidation; or
 - (c) make it a condition of the supply of the service to a company in liquidation that the liquidator personally guarantees payment of the charges that would be incurred for the supply of the service.
- (2) The charges incurred by a liquidator for the supply of an essential service are an expense incurred by the liquidator for the purposes of clause 15(a) of Schedule 8.

Subdivision G Liquidators' Rights to Company's Documents

29 Liquidator may require director, etc. to deliver documents

A liquidator may, by notice in writing, require a director or shareholder of the company or any other person to deliver to the liquidator any records or documents of the company in that person's possession or under that person's control as the liquidator requires.

30 Liquidator may require director, etc. to provide information

- (1) A liquidator may, from time to time, by notice in writing, require the following persons to do any of the things specified in subsection (2):
 - (a) a director or former director of the company;
 - (b) a shareholder of the company;

- (c) a person who was involved in the promotion or formation of the company;
- (d) a person who is, or has been, an employee of the company;
- (e) a receiver, accountant, auditor, bank officer, or other person having knowledge of the affairs of the company;
- (f) a person who is acting, or who has at any time acted, as legal practitioner for the company.
- (2) A person referred to in subsection (1) may be required to:
 - (a) attend on the liquidator at such reasonable time or times and at such place as may be specified in the notice;
 - (b) provide the liquidator with such information about the business, accounts, or affairs of the company as the liquidator requests;
 - (c) be examined on oath or affirmation by the liquidator or by a legal practitioner acting on behalf of the liquidator on any matter relating to the business, accounts, or affairs of the company;
 - (d) assist in the liquidation to the best of the person's ability.
- (3) Without limiting subsection (2)(a), a person may be required to attend on the liquidator under that subsection at a meeting of creditors of the company.

31 Reasonable expenses may be paid

- (1) Without limiting subsection (2), the liquidator may pay to a person referred to in section 30(1)(d), (e), or (f), who is not an employee of the company, reasonable travelling and other expenses in complying with a requirement of the liquidator under that section.
- (2) The Court may, on the application of the liquidator or a person referred to in section 30(1)(d), (e), or (f), who is not an employee of the company, order that that person is entitled to receive reasonable remuneration and travelling and other expenses in complying with a requirement of the liquidator under that section.
- (3) A person referred to in section 30(1)(d), (e), or (f) is not entitled to

refuse to comply with a requirement of the liquidator under that section by reason only that:

- (a) an application to the Court to be paid remuneration or travelling and other expenses has not been made or determined; or
- (b) remuneration or travelling and other expenses to which that person is entitled have not been paid in advance; or
- (c) the liquidator has not paid that person travelling or other expenses.

32 Examination by liquidator

- (1) A liquidator, or legal practitioner acting on behalf of the liquidator, may administer an oath to, or take the affirmation of, a person required to be examined under section 30.
- (2) A person required to be examined under section 30 is entitled to be represented by a legal practitioner.
- (3) A liquidator, or legal practitioner acting on behalf of the liquidator, who conducts an examination under section 30 must ensure that the examination is recorded in writing or by means of a sound recording, video and sound recording, or other similar means.

33 Court may order person to comply with section 30

- (1) The Court may, on the application of the liquidator, order a person who has failed to comply with a requirement of the liquidator under section 30 to comply with that requirement.
- (2) The Court may, on the application of the liquidator, order a person to whom section 30 applies to:
 - (a) attend before the Court and be examined on oath or affirmation by the Court or the liquidator, or legal practitioner acting on behalf of the liquidator, on any matter relating to the business, accounts, or affairs of the company;
 - (b) produce any documents relating to the business, accounts, or affairs of the company in that person's possession or under that person's control.

- (3) If a person is examined under subsection (2)(a):
 - (a) the examination must be recorded in writing; and
 - (b) the person examined must sign the record.
- (4) Subject to any directions by the Court, a record of an examination under this section is admissible in evidence in any proceedings under section 77 of the *Companies Act 2009* or this Division.

34 Self-incrimination no excuse

- (1) A person is not excused from answering a question in the course of being examined under section 30 or 33 on the ground that the answer may incriminate or tend to incriminate that person.
- (2) The testimony of the person examined is not admissible as evidence in criminal proceedings against that person except on a charge of perjury in relation to that testimony.

35 Restriction on enforcement of lien over company's documents

- (1) A person is not entitled, as against the liquidator of a company, to claim or enforce a lien over documents of the company.
- (2) If the lien arises in relation to a debt for the provision of services to the company before the commencement of the liquidation, the debt is a preferential claim against the company under section 50 to the extent of \$500 or such greater amount as may be prescribed at the commencement of the liquidation.
- (3) Nothing in this section applies to a company that was put into liquidation under section 16 or 17 if:
 - (a) the directors of the company passed a resolution of the kind referred to in section 42; and
 - (b) section 43 does not apply in relation to the company.

36 Delivery of document held by secured creditor

(1) A person is required to deliver a document to a liquidator under section 29 even though possession of the document creates a

- security interest over property of a company.
- (2) Production of the document to the liquidator does not prejudice the existence or priority of the security interest, and the liquidator must make the document available to the person entitled to it for the purpose of dealing with, or realising the security interest over, the secured property.

37 Documents held by receiver

- (1) A receiver is not required to deliver to a liquidator any documents that the receiver requires for the purpose of exercising any powers or functions as receiver in relation to property of a company in liquidation.
- (2) The liquidator may, from time to time, by notice in writing, require the receiver:
 - (a) to make any records and documents available for inspection by the liquidator at any reasonable time or times; and
 - (b) to provide the liquidator with copies of any records and documents or extracts from them.
- (3) The liquidator may take copies of any records and documents made available for inspection or extracts from them.
- (4) The liquidator must pay the reasonable expenses of the receiver in complying with a requirement of the liquidator under subsection (2).

Subdivision H Meetings

38 Notice of first creditors' meeting

- (1) A liquidator must give to every known creditor a notice in writing of a meeting of creditors, and:
 - (a) if section 47(1) applies, the notice must be given together with the report and notice referred to in that subsection; and
 - (b) if the liquidator receives a notice under section 44(1)(c) requiring a meeting of creditors to be called, the notice must be given within 10 working days after receiving the notice.

(2) Not less than 5 working days before the creditors' meeting, a liquidator must also give public notice of the meeting.

39 Timing of first creditors' meeting

- (1) Except if section 38(1)(b) applies, a meeting of creditors must be held:
 - (a) in the case of a liquidator appointed under section 16 or 17, within 10 working days of the liquidator's appointment; or
 - (b) in the case of a liquidator appointed under section 18, within 30 working days of the liquidator's appointment; or
 - (c) in either case, within such longer period as the Court may allow.
- (2) If section 38(1)(b) applies, a meeting of creditors must be held within 15 working days after the liquidator receives a notice under section 44(1)(c) requiring a meeting of creditors to be called.

40 Purpose of first creditors' meeting

- (1) Subject to sections 42 and 44, the liquidator of a company must call a meeting of the creditors of the company for the purpose:
 - in the case of a liquidator appointed under section 16 or 17, of resolving whether to confirm the appointment of that liquidator or to appoint another liquidator in place of the liquidator so appointed; or
 - (b) in the case of a liquidator appointed under section 18, of resolving whether to confirm the appointment of that liquidator or to make an application to the Court for the appointment of a liquidator in place of the liquidator so appointed; or
 - (c) in either case, of determining whether to pass a resolution for the purposes of section 46(1)(c).
- (2) If the appointment of a liquidator under section 16 or 17 is not confirmed at a meeting of creditors and another liquidator is not appointed in place of that liquidator, the appointment of the liquidator under either of those sections continues until another liquidator is appointed.

41 Replacement liquidator

- (1) If at a meeting of creditors it is resolved to appoint a person as liquidator of the company in place of the liquidator appointed under section 16 or 17, the person who it is resolved to appoint as liquidator is, subject to section 15, appointed as the liquidator of the company.
- (2) If at a meeting of creditors it is resolved to apply to the Court for the appointment of a person as liquidator in place of the liquidator appointed under section 18:
 - (a) the liquidator of the company must immediately apply to the Court for the appointment of that person as liquidator; and
 - (b) the Court may appoint that person as the liquidator of the company.

42 Effect of directors' resolving company able to pay its debts

Nothing in sections 38 to 41 applies to the liquidator of a company appointed under section 16 or 17 if, within 20 working days before the appointment of the liquidator, the directors of the company resolved that the company would, on the appointment of a liquidator under either of those sections, be able to pay its debts and a copy of the resolution is delivered to the Registrar for registration.

43 Other creditors' meetings

- (1) Subject to section 44, the liquidator of a company who was not, by reason of section 38, required to call a meeting of creditors of the company, must immediately call a meeting of the creditors of the company for the purpose specified in section 40(1)(a) or (b), if the liquidator is satisfied that:
 - (a) the directors who voted in favour of a resolution referred to in that subsection did not have reasonable grounds to believe that the company would, on the appointment of a liquidator under section 16 or 17, be able to pay its debts; or
 - (b) the company is not able to pay its debts.
- (2) Section 40 applies with the necessary modifications.

44 Liquidator may dispense with meetings of creditors

- (1) A liquidator is not required to call a meeting of creditors under section 38 or 43, as the case may be, if:
 - (a) the liquidator considers that no such meeting should be held, having regard to:
 - (i) the assets and liabilities of the company; and
 - (ii) the likely result of the liquidation of the company; and
 - (iii) any other relevant matters; and
 - (b) the liquidator gives notice in writing to the creditors stating:
 - (i) that the liquidator does not consider that a meeting should be held; and
 - (ii) the reasons for the liquidator's view; and
 - (iii) that no such meeting will be called unless a creditor gives notice in writing to the liquidator, within 10 working days after receiving the notice, requesting a meeting to be called and giving reasons why a meeting should be called; and
 - (c) no notice requesting a meeting to be called is received by the liquidator within that period or, if a notice requesting a meeting to be called is received within that period, the Court directs the liquidator that, having regard to the reasons given in the notice and the circumstances of the company considered by the liquidator under paragraph (a), it is not necessary for the liquidator to call a meeting.
- (2) Notice under subsection (1)(b) must be given to every known creditor:
 - (a) if section 45(1) applies, together with the report and notice referred to in that section; or
 - (b) if section 45(1) is not applicable, at the time the liquidator would have been required to send the report and notice referred to in

that section if it were applicable.

45 Meetings of creditors or shareholders

- (1) At any time in the course of the liquidation, the liquidator may, at the request in writing of any creditor or shareholder or on the liquidator's own motion, call a meeting of creditors or shareholders:
 - (a) to vote on a proposal that a liquidation committee be appointed to act with the liquidator in accordance with Schedule 6; and
 - (b) if it is so decided, to choose the members of the committee.
- (2) A liquidator may decline a request by a creditor or shareholder to call a meeting on the ground that:
 - (a) the request is frivolous or vexatious; or
 - (b) the request was not made in good faith; or
 - (c) except if a creditor or shareholder agrees to meet the costs, the costs of calling a meeting would be out of all proportion to the value of the company's assets.
- (3) The decision of a liquidator to decline the request may be reviewed by the Court on the application of any creditor or shareholder, as the case may be.
- (4) Subject to subsections (2) and (3), a liquidator who receives a request to call:
 - (a) a meeting of creditors, must immediately call a meeting in accordance with Schedule 2; or
 - (b) a meeting of shareholders, must immediately call a meeting in accordance with the company's rules, except the liquidator has power to give notice of a meeting of shareholders and to act as, or appoint, the chairperson of the meeting.
- (5) The sole shareholder of a company may present to the liquidator a view on any matter that could have been decided at a meeting of shareholders under this section, and that view must, for all purposes, be treated as though it were a decision taken at a meeting of

shareholders.

46 Views of creditors and shareholders at meetings to be considered

- (1) The liquidator must consider the views of the following:
 - (a) the shareholders by whom any special resolution was passed at a meeting held for the purposes of section 17 set out in a resolution passed at that meeting;
 - (b) creditors set out in any resolution passed at a meeting held for the purposes of section 40;
 - (c) creditors or shareholders set out in a resolution passed at a meeting called in accordance with subsection (2);
 - (d) any liquidation committee given in writing to the liquidator.
- (2) For the purposes of subsection (1), a liquidator:
 - (a) must summon meetings of shareholders at such times as may be specified by any resolution of shareholders passed at a meeting held for the purposes of section 17; or
 - (b) must summon meetings of creditors at any times that may be specified by any resolution of creditors passed at a meeting held for the purposes of section 40; or
 - (c) must summon a meeting of shareholders immediately when required to do so by notice in writing given by shareholders holding not less than 10% of all shares issued by the company; or
 - (d) must summon a meeting of creditors immediately when required to do so by notice in writing given by creditors to whom is owed not less than 10% of the total amount owed to all creditors of the company; or
 - (e) may summon a meeting of shareholders or creditors of the company.
- (3) A liquidator who calls a meeting of creditors or shareholders must

call a meeting:

- (a) in accordance with the rules of the company, in the case of a meeting of shareholders; or
- (b) in accordance with Schedule 2, in the case of a meeting of creditors.
- (4) For the purposes of holding a meeting of shareholders under subsection (3)(a), the liquidator is deemed to have power under the company's rules to call a meeting of shareholders despite anything in the company's rules, and references in the company's rules to chairman or chairperson must be read as references to the liquidator.
- (5) Nothing in this section limits or prevents a liquidator from exercising his or her discretion in carrying out his or her functions and duties under this Act.

Subdivision I Reports

47 First report

- (1) A liquidator must, within the applicable period:
 - (a) prepare a list of every known creditor of the company; and
 - (b) prepare and send to every known creditor, every shareholder, and the Registrar for registration:
 - (i) a report containing a statement of the company's affairs, proposals for conducting the liquidation, and, if practicable, the estimated date of its completion; and
 - (ii) a notice explaining the right of a creditor or shareholder to require the liquidator to call a meeting of creditors or shareholders (as the case may be) under section 45.
- (2) For the purposes of subsection (1), "applicable period" means:
 - (a) in the case of a liquidator appointed under section 16, or 17, 45 working days after the liquidator's appointment; or
 - (b) in the case of a liquidator appointed under section 18, 25

24

Reprint authorised by the Attorney General and published under the Legislation Act 2023

working days after the liquidator's appointment; or

(c) in either case, any longer period that the Court may allow.

48 Six-monthly report

A liquidator must, within 2 months of the end of each period of 6 months following the date of commencement of the liquidation, prepare and send to every known creditor and every shareholder, and send or deliver to the Registrar, a report:

- (a) on the conduct of the liquidation during the preceding 6 months; and
- (b) of any further proposals that the liquidator has for completing the liquidation.

49 Exemption from reporting requirements

- (1) The Court may, on the application of a liquidator and on any conditions that the Court thinks fit:
 - (a) exempt the liquidator from compliance with section 47 or 48; or
 - (b) modify the application of those sections in relation to the liquidator.
- (2) The liquidator need not comply with section 47 or 48 if the liquidator is satisfied that the value of the assets of the company available for distribution to unsecured creditors who are not preferential creditors is not likely to exceed \$0.20, or any other prescribed sum, in every dollar owed to those creditors.
- (3) If subsection (2) applies, and the liquidator does not intend to comply with section 47 or 48, the liquidator must give notice to the Registrar that he or she does not intend to comply with those sections.

Subdivision J Creditors' Claims

50 Preferential claims

(1) The liquidator must pay out of the assets of the company the expenses, fees, and claims set out in Part 3 of Schedule 8 to the

25

extent and in the order of priority specified in that schedule.

(2) Without limiting clause 18 of Schedule 8, assets in subsection (1) does not include assets subject to a security interest unless the security interest is surrendered or taken to be surrendered or redeemed under Part 2 of that Schedule.

51 Claims of other creditors and distribution of surplus assets

- (1) After paying preferential claims in accordance with section 50, the liquidator must apply the assets of the company in satisfaction of all other claims.
- (2) The claims referred to in subsection (1) rank equally among themselves and must be paid in full, unless the assets are insufficient to meet them, in which case payment abates rateably among all claims.
- (3) If, before the commencement of a liquidation, a creditor agrees to accept a lower priority in respect of a debt than that which it would otherwise have under this section, nothing in this section prevents the agreement from having effect according to its terms.
- (4) A person whose shares have been repurchased by the company, or a person whose shares have been redeemed by the company, but who has not received payment in full of the repurchase price or redemption price (as the case may be) is taken to have agreed to subordinate his or her claim for any unpaid balance of the repurchase price or redemption price (as the case may be) to the rights of other creditors of the company.
- (5) Subject to section 58 and to clause 25 of Schedule 8, after paying the claims referred to in subsections (1) and (3), the liquidator must distribute the company's surplus assets:
 - (a) in accordance with the company's rules; or
 - (b) if the company's rules do not provide for the distribution of surplus assets, in accordance with section 22(1)(c) of the *Companies Act 2009* and any preferential rights as to distributions of capital attached to shares issued by the company in accordance with section 22(3)(b) of the *Companies*

Act 2009.

Subdivision K End of Liquidation

52 Completion of liquidation

The liquidation of a company is completed when the liquidator:

- (a) complies with section 53(2); or
- (b) delivers to the Registrar for registration:
 - (i) a copy of any order made by the Court under section 53(3)(a); or
 - (ii) a copy of any order made by the Court under section 53(3)(b) together with any documents required to comply with the order.

Final report and accounts

- (1) As soon as practicable after completing his or her duties in relation to the liquidation, the liquidator of a company must prepare and send to every creditor whose claim has been admitted and to every shareholder:
 - (a) the final report and statement of realisation and distribution in respect of the liquidation; and
 - (b) a statement that:
 - (i) all known assets have been disclaimed, or realised, or distributed without realisation; and
 - (ii) all proceeds of realisation have been distributed; and
 - (iii) the company is ready to be removed from the Solomon Islands register; and
 - (c) a summary of the applicable grounds on which the creditor or shareholder may object to the removal of the company from the Solomon Islands register under section 148 of the *Companies* Act 2009.

- (2) As soon as practicable after completing his or her duties in relation to the liquidation, the liquidator of a company must send or deliver copies of the documents referred to in subsection (1) to the Registrar for registration.
- (3) The Court may, on the application of a liquidator and on any conditions that the Court thinks fit:
 - (a) exempt the liquidator from compliance with subsection (1) or (2); or
 - (b) modify the application of those provisions in relation to the liquidator.

54 Liquidation surplus account

- (1) Money representing unclaimed assets of a company standing to the credit of a liquidator must, after completion of the liquidation, be paid to the Registrar to be held on trust and dealt with in accordance with this section.
- (2) At the expiration of a period of 12 months after the date on which the money is paid, the Registrar must, after deduction of any amount required to meet the claim of any person that is established within that period, pay the balance into an account entitled the "Liquidation Surplus Account" for distribution in accordance with this section.
- (3) Money held in the Liquidation Surplus Account may be invested in accordance with the law as to the investment of trust funds by trustees. Interest on any investment must be distributed in accordance with this section.
- (4) Money held in the Liquidation Surplus Account may be:
 - (a) paid or distributed to any person entitled to payment or distribution in the liquidation of a company any money representing the surplus assets of which has been credited to the Liquidation Surplus Account; or
 - (b) paid, subject to such conditions as the Registrar may impose, in meeting costs incurred in the course of liquidation of a company for the purpose of proceedings brought by the

company, including legal or other expert advice, or the costs of any expert witness, if the Registrar is satisfied that it is fair and reasonable for those costs to be met out of the Liquidation Surplus Account.

(5) In making a payment under this section, the Registrar is not required to ascertain that money or sufficient money was received on account of any company to which the claim for payment relates.

55 Termination of liquidation by Court

- (1) The Court may, at any time after the appointment of a liquidator of a company, if it is satisfied that it is just and equitable to do so, make an order terminating the liquidation of the company.
- (2) An application may be made by:
 - (a) the liquidator of the company; or
 - (b) a director of the company; or
 - (c) a shareholder of the company; or
 - (d) a creditor of the company; or
 - (e) the Registrar.
- (3) The Court may require the liquidator of the company to give a report to the Court with respect to any facts or matters relevant to the application.
- (4) If the Court makes an order, the company ceases to be in liquidation and the liquidator ceases to hold office with effect on and from the making of the order or any other date specified in the order.
- (5) The Court may, on, or at any time after, making an order, make any other order that it thinks fit in connection with the termination of the liquidation.

Notice of termination of liquidation

(1) The person who applied for a Court order terminating the liquidation or the liquidator, in the case of the creditors terminating the

liquidation, must, within 10 working days after the order was made, or the resolution was passed, (as the case may be), deliver a notice of the order, or the passing of the resolution, to the Registrar for registration.

(2) A person who fails to comply with subsection (1) commits an offence and is liable on conviction to a fine not exceeding 2,500 penalty units.

57 Liquidation of assets of overseas company in Solomon Islands

- (1) An application may be made to the Court for the liquidation of the assets in Solomon Islands of an overseas company in accordance with this Part subject to the modifications and exclusions set out in Schedule 9.
- (2) An application may be made under subsection (1) whether or not the overseas company:
 - (a) is registered under the *Companies Act 2009*; or
 - (b) has given public notice of an intention to cease to carry on business in Solomon Islands; or
 - (c) has given notice to the Registrar of the date on which it will cease to carry on business in Solomon Islands; or
 - (d) has been dissolved, or otherwise ceased to exist as a company, under or by virtue of the laws of any other country.

Liquidation of assets of community companies

- (1) If any surplus assets remain after all payments and claims are made under this Part the surplus assets must be distributed in proportion to the number of the shares held by each shareholder and held by such shareholders on trust for the benefit of the community.
- (2) An application may be made to the Court by a member of the community which is the beneficiary of the community company for the residual assets to be held by a Court appointed person.
- (3) The Court when considering any application made under subsection (2) must consider the interests of the community and the community purpose of the company.

PART 3 RECEIVERSHIPS

Division 1 Application

59 Application of this Part to receivers

- (1) This Part applies:
 - (a) to a receiver appointed after the commencement of this Act; and
 - (b) with the exceptions and modifications specified in section 60, to a receiver holding office on the commencement of this Act.
- (2) Except as expressly provided in this Act, Schedules 10 and 11 apply to receivers.

60 Application of this Part to receivers

In the application of this Part to a receiver holding office on the commencement of this Act:

- (a) clause 1 of Schedule 11 does not apply;
- (b) section 68 applies if the reference in that section to the receivers appointment were a reference to the commencement of this Act unless the receiver has under Part VI of the Companies Act (Cap. 175) already prepared a statement, in which case section 68 does not apply;
- (c) section 70 applies as if the reference in that section to the receiver's appointment were a reference to the commencement of this Act:
- (d) section 70 (2) does not apply to a receivership that ended before the commencement of this Act, in which case the provisions of Part VI of the *Companies Act* (Cap. 175) continue to apply;
- (e) section 78 does not apply to a receivership that ended before the commencement of this Act;

(f) clauses 19(1)(b) and (c) and 20(1) and (2) of Schedule 10 do not apply.

Division 2 Appointment of Receivers

Restrictions on appointment of receiver

- (1) The following persons must not be appointed as a receiver:
 - (a) a person who must not be appointed as a receiver under clause 1 of Schedule 11, unless the Court orders otherwise;
 - (b) a person who has not consented in writing to their appointment as a receiver.
- (2) A person who fails to comply with subsection (1)(a) commits an offence and is liable on conviction to a fine not exceeding 2,500 penalty units.

Appointment of receiver under document

- A receiver may be appointed in respect of the property of a person by, or in the exercise of a power conferred by, a document to which the person is a party.
- (2) The appointment of a receiver in the exercise of a power conferred by a document must be in writing.
- (3) The power conferred by subsection (1) is in addition to the power conferred on the Court to appoint a receiver.

Extent of power to appoint receiver under document

A power conferred by a document to appoint a receiver includes, unless the document expressly provides otherwise, the power to appoint:

- (a) 2 or more receivers; or
- (b) a receiver additional to 1 or more presently in office; or
- (c) a receiver to succeed a receiver whose office has become vacant.

Division 3 Notices

Notices given by receiver

- (1) Not later than 5 working days after being appointed, a receiver must:
 - (a) give written notice of his or her appointment to the grantor; and
 - (b) give public notice of his or her appointment, including:
 - (i) the receiver's full name; and
 - (ii) the date of the appointment; and
 - (iii) the receiver's office address; and
 - (iv) a brief description of the property in receivership.
- (2) If the grantor is a body corporate, the receiver must, within 5 working days after being appointed, send a copy of the public notice to the Registrar.
- (3) If the appointment of the receiver is in addition to a receiver who already holds office or is in place of a person who has vacated office as receiver, as the case may be, every notice under this section must state that fact.
- (4) A receiver who fails to comply with this section commits an offence and is liable on conviction to a fine not exceeding 2,500 penalty units.

65 Notice of receivership

- (1) Any document provided by or on behalf of a grantor or a receiver in the course of a receivership must clearly state that a receiver has been appointed in respect of the property that is in receivership.
- (2) A failure to comply with subsection (1) does not affect the validity of the document.
- (3) A person who fails to comply with subsection (1) commits an offence and is liable on conviction to a fine not exceeding 1,250 penalty units.

Division 4 Obligations to Receivers

66 Obligations of grantor

- (1) A grantor and, in the case of a grantor that is a body corporate, every director of the grantor, must:
 - (a) make available to the receiver all documents and information relating to the property in receivership in the grantor's possession or under the grantor's control;
 - (b) if required to do so by the receiver, verify by statutory declaration that the documents and information are complete and correct:
 - (c) give the receiver any assistance that the receiver may reasonably require;
 - (d) if the grantor has a common seal, make the common seal available for use by the receiver.
- (2) On the application of the receiver, the Court may make an order requiring the grantor or, if the grantor is a body corporate, a director of the grantor to comply with subsection (1).

67 Obligations of suppliers of essential services

Despite any other written law or any contract, a supplier of an essential service must not:

- (a) refuse to supply the service to a receiver or to the owner of property in receivership by reason of the grantor's default in paying charges due for the service in relation to a period before the date of the appointment of the receiver; or
- (b) make it a condition of the further supply of the service to a receiver or to the owner of property in receivership that payment be made of outstanding charges due for the service in relation to a period before the date of the appointment of the receiver.

Division 5 Reports of Receivers

68 First report of receiver

- (1) Not later than 3 months after the receiver's appointment, a receiver must:
 - (a) prepare a report on the state of affairs with respect to the property in receivership; and
 - (b) send a copy of the report to the grantor and any person in whose interests the receiver was appointed; and
 - (c) in the case of a receiver appointed by the Court, file a copy of the report in the office of the Court; and
 - (d) in the case of a grantor that is a body corporate, send a copy of the report to the Registrar.
- (2) A receiver who fails to comply with this section or section 69 commits an offence and is liable on conviction to a fine not exceeding 2,500 penalty units.

69 Contents of first report

A receiver's first report must include details of:

- (a) the events leading up to the appointment of the receiver, so far as the receiver is aware of them; and
- (b) any property disposed of and any proposals for the disposal of property in receivership; and
- (c) any amounts owing, as at the date of appointment, to any person in whose interests the receiver was appointed; and
- (d) any amounts owing, as at the date of appointment, to creditors of the grantor having preferential claims; and
- (e) the receiver's expenses and remuneration or likely expenses and remuneration; and
- (f) any amounts likely to be available for payment to creditors other

- than those referred to in paragraph (c) or (d); and
- (g) the assets comprising the property in receivership; and
- (h) the debts and liabilities to be satisfied from the property in receivership; and
- (i) the names and addresses of the creditors with an interest in the property in receivership; and
- (j) any encumbrance over the property in receivership held by any creditor including the date on which it was created; and
- (k) any default by the grantor in making relevant information available.

70 Further reports of receiver

- (1) Not later than 2 months after the end of each period of 6 months after the receiver's appointment, a receiver must:
 - (a) prepare a further report summarising:
 - (i) the state of affairs with respect to the property in receivership as at that date; and
 - (ii) the conduct of the receivership, including all amounts received and paid, during the period to which the report relates: and
 - (b) send a copy of the report to the grantor and any person in whose interests the receiver was appointed; and
 - (c) in the case of a receiver appointed by the Court, file a copy of the report in the office of the Court; and
 - (d) in the case of a grantor that is a body corporate, send a copy of the report to the Registrar.
- (2) Not later than 2 months after the date on which the receivership ends, a person who was a receiver at the end of the receivership must:
 - (a) prepare a further report summarising:

- (i) the state of affairs with respect to the property in receivership as at that date; and
- (ii) the conduct of the receivership, including all amounts received and paid, during the period to which the report relates; and
- (b) send a copy of the report to the grantor and any person in whose interests the receiver was appointed; and
- (c) in the case of a receiver appointed by the Court, file a copy of the report in the office of the Court; and
- (d) in the case of a grantor that is a body corporate, send a copy of the report to the Registrar.
- (3) A person who fails to comply with this section or section 71 commits an offence and is liable on conviction to a fine not exceeding 2,500 penalty units.

71 Contents of further reports

Any report under section 70 must include details of:

- any property disposed of since the date of any previous report and any proposals for the disposal of property in receivership; and
- (b) any amounts owing, as at the date of the report, to any person in whose interests the receiver was appointed; and
- (c) any amounts owing, as at the date of the report, to creditors of the grantor having preferential claims; and
- (d) the receiver's expenses and remuneration or likely expenses and remuneration; and
- (e) any amounts likely to be available as at the date of the report for payment to creditors other than those referred to in paragraph (b) or (c).

72 What may be omitted from reports

A receiver may omit from any report under this Part details of any proposals for disposal of property in receivership that the receiver considers would materially prejudice the carrying out of the receivership by the receiver.

73 Extension of time for preparing reports

A period of time within which a receiver is required to prepare a report under this Part may be extended, on the application of that person, by:

- (a) the Court, if the person was appointed a receiver by the Court;
- (b) the Registrar, if the person was appointed a receiver by or under a document.

74 Persons entitled to receive reports

- (1) A receiver who is requested in writing by a person referred to in subsection (2) to provide a copy of a report under this Part must, not later than 5 working days after receiving the request, send a copy of the report to that person.
- (2) The persons referred to in subsection (1) are a creditor, director, or guarantor of the grantor, or any other person with an interest in any of the property in receivership, who has paid the reasonable costs of making and sending a copy of the requested report.
- (3) A person who fails to comply with this section commits an offence and is liable on conviction to a fine not exceeding 2,500 penalty units.

75 Persons entitled to inspect reports

A person to whom a report must be sent in accordance with section 74 is entitled to inspect the report during normal office hours at the office of the receiver.

Division 6 Preferential Claims

76 Preferential claims

- (1) This section applies to a receiver of the property of a grantor that is a company, other than a company in liquidation at the time of the receiver's appointment, and who was appointed under a security agreement that created or provided for a security interest that:
 - (a) is over all or any part of the company's accounts receivable and inventory or all or any part of either of them; and
 - (b) is not a purchase money security interest that has been perfected at the time specified in section 21 of the Secured Transactions Act 2008; and
 - (c) is not a security interest that has been perfected under the Secured Transactions Act 2008 at the time of the receiver's appointment and that arises from the transfer of an account receivable for which new value is provided by the transferee for the acquisition of that account receivable (whether or not the transfer of the account receivable secures payment or performance of an obligation).
- (2) A receiver to whom this section applies must apply accounts receivable and inventory that are subject to the security interest or their proceeds:
 - (a) first, to reimburse the receiver for his or her expenses and remuneration; and
 - (b) secondly, to pay the claims of a person who has:
 - (i) a purchase money security interest over all or any of those assets, that have been perfected at the time specified in section 21 of the Secured Transactions Act 2008:
 - (ii) a security interest over all or any of those assets, that have been perfected under the Secured Transactions Act 2008 at the time of the receivers appointment and that arises from the transfer of an account receivable for which

new value is provided by the transferee for the acquisition of that account receivable (whether or not the transfer of the account receivable secures payment or performance of an obligation); and

- (c) thirdly to pay preferential claims to the extent and the order of priority specified in Schedule 8.
- (3) The receiver must apply the accounts receivable and inventory as set out in subsection (2) before paying the claims of any person under a security interest, other than a security interest referred to in subsection (2)(b).
- (4) In the application of Schedule 8 in accordance with subsection (2):
 - (a) references to a liquidator are to be read as references to a receiver; or
 - (b) references to the commencement of the liquidation are to be read as references to the appointment of the receiver; or
 - (c) reference to a company being put into or being in liquidation are to be read as references to the company being put into or being in receivership.

77 Receiver appointed under floating security interest: ranking of preferential claims

The provisions of section 76 do not apply in relation to a grantor in respect of whose property a receiver was appointed under a floating security interest before the commencement of this Act and the provisions of the *Companies Act* (Cap. 175) shall continue to apply in relation to that grantor notwithstanding the repeal of that Act by this Act.

Division 7 End of Receivership

78 Notice of end of receivership

(1) If the grantor is a body corporate, the person who held office as receiver at the end of the receivership must, not later than 5 working days after the receivership ends, send or deliver to the Registrar notice in writing of the fact that the receivership has ended.

(2) A person who fails to comply with subsection (1) commits an offence and is liable on conviction to a fine not exceeding 2,500 penalty units.

79 Court may terminate or limit receivership

- (1) The Court may, on the application of the grantor or a liquidator of the grantor:
 - (a) order that a receiver cease to act as such as from a specified date, and prohibit the appointment of any other receiver in respect of the property in receivership;
 - (b) order that a receiver, as from a specified date, act only in respect of specified assets forming part of the property in receivership.
- (2) An order may be made only if the Court is satisfied that:
 - (a) the purpose of the receivership has been satisfied so far as possible; or
 - (b) circumstances no longer justify continuation of the receivership.
- (3) Unless the Court orders otherwise:
 - (a) a copy of an application under this section must be served on the receiver not less than 5 working days before the hearing of the application; and
 - (b) the receiver may appear and be heard at the hearing.
- (4) An order under subsection (1) may be made on any conditions that the Court thinks fit.
- (5) An order under this section does not affect a security or security interest over the property in respect of which the order is made.
- (6) The Court may, on the application of any person who applied for or is affected by the order, rescind, or amend an order made under this section.

PART 4 MISCELLANEOUS

81 Regulations

- (1) The Minister may make regulations to give effect to the provisions of this Act, and in particular for all or any of the following purposes:
 - (a) prescribing fees or other amounts payable to the Registrar in respect of any matter under this Act;
 - (b) prescribing fees or other amounts payable to the Registrar of the High Court in respect of any Court proceedings under this Act:
 - (c) prescribing forms (including Court forms) for the purposes of this Act; and those regulations may require:
 - (i) the inclusion in, or attachment to, forms of specified information or documents;
 - (ii) forms to be signed by specified persons;
 - (d) providing for and regulating any Court proceedings under this Act:
 - (e) prescribing transitional and savings provisions relating to the coming into force of this Act;
 - (f) providing for any matters contemplated by this Act, necessary for its administration, or necessary for giving it full effect.
- (2) The Registrar or the Registrar of the High Court, as the case may require, may refuse to perform a function or exercise a power until the prescribed fee or amount is paid.
- (3) Any regulations made under subsection (1) may authorise the Registrar or the Registrar of the High Court, as the case may require, to waive, in whole or in part and on any conditions that may be prescribed, payment of any amount referred to in paragraphs (a) or (b) of that subsection.
- (4) Any fee or amount payable to the Registrar or to the Registrar of the High Court, as the case may require, is recoverable by the Registrar

or the Registrar of the High Court, as the case may require, in any court of competent jurisdiction as a debt due to the Crown.

(Section 3)

INTERPRETATION

1 Definitions

In this Act, unless the context otherwise requires:

"accounting period", in relation to a company, means a year ending on a balance date of the company and, if as a result of the date of the registration of the company or a change of the balance date of the company, the period ending on that date is longer or shorter than a year, that longer or shorter period is an accounting period;

"arrangement" includes a reorganisation of the share capital of a company by the consolidation of shares of different classes, or by the division of shares into shares of different classes, or by both those methods;

"balance date", in relation to a company, means the close of 31 March or of any other date that the directors of the company adopt as the company's balance date in accordance with any regulations made under this Act;

"board and board of directors", in relation to a company, means:

- (a) directors of the company who number not less than the required quorum acting together as a board of directors; or
- (b) if the company has only 1 director, that director;

"broadcasting" means any transmission of programmes, whether or not encrypted, by radio waves or other means of telecommunication for reception by the public by means of broadcasting receiving apparatus; but does not include any such transmission of programmes:

(a) made on the demand of a particular person for reception only by that person; or

44

(b) made solely for performance or display in a public place;

"company" means a company registered or reregistered under the Companies Act 2009;

"compromise" means a compromise between a company and its creditors, including a compromise:

- (a) cancelling all or part of a debt of the company; or
- (b) varying the rights of its creditors or the terms of a debt; or
- (c) relating to an alteration of a company's rules that affects the likelihood of the company being able to pay a debt;

"Court" means the High Court of Solomon Islands;

"court officer" means the Registrar or other officer of the Court;

"creditor":

- (a) in Division 1 of Part 2 (compromises with creditors) includes:
 - a person who, in a liquidation, would be entitled to claim that a debt is owing to that person by the company; and
 - (ii) a secured creditor;
- (b) in Division 2 of Part 2 (liquidations):
 - (i) means a person who, in a liquidation, would be entitled to claim in accordance with clause 2 of Schedule 8 that a debt is owing to that person by the company; and
 - (ii) includes a secured creditor only:
 - (A) for the purposes of sections 18, 22, and 55 and clause 5 of Schedule 5; or
 - (B) to the extent of the amount of any debt owing

to the secured creditor in respect of which the secured creditor claims under Part 2 of Schedule 8 as an unsecured creditor;

"creditor" includes a person to whom the grantor owes a debt or is under a liability, whether present or future, certain or contingent, and whether an ascertained debt or liability or a liability in damages;

"director", in relation to a company:

- (a) includes a person occupying the position of director of the company by whatever name called; but
- (b) does not include a receiver;

"directors" has the same meaning as the definitions of "board" and "board of directors":

"distribution", in relation to a distribution by a company to a shareholder, means:

- (a) the direct or indirect transfer of money or property, other than the company's own shares, to or for the benefit of the shareholder; or
- (b) the incurring of a debt to or for the benefit of the shareholder,

in relation to shares held by that shareholder, whether by means of a dividend, a purchase, redemption or other acquisition of shares, a distribution of indebtedness, or some other means;

"document":

- (a) means information in written or electronic form, or both; and
- (b) includes anything from which information may be reproduced (with or without the aid of anything else);

"electronic" includes electrical, digital, magnetic, optical, electromagnetic, biometric, and photonic;

"enforcement process", in relation to property, means:

- (a) execution against that property; or
- (b) any other enforcement process in relation to that property that involves a court:

"essential service" means:

- (a) the retail supply of electricity;
- (b) the retail supply of fuel and other similar consumable items necessary for the generation of electricity;
- (c) the retail supply of gas;
- (d) the supply of water;
- (e) telecommunications services;

"financial statements", in relation to a company and a balance date, means:

- (a) a statement of financial position for the company as at the balance date; and
- (b) in the case of:
 - (i) a company trading for profit, a statement of financial performance for the company in relation to the accounting period ending at the balance date; and
 - (ii) a company not trading for profit, an income and expenditure statement for the company in relation to the accounting period ending at the balance date; and
- (c) if required by regulations made under this Act, a statement of cash flows for the company in relation to the accounting period ending on the balance date; and
- (d) any other financial statements in relation to the company or any group of companies of which it is the holding

- company as may be required by regulations made under this Act; and
- (e) any notes or documents giving information relating to the statement of financial position and other statements;

"grantor" means the person in respect of whose property a receiver is, or may be, appointed;

"holding company" means a holding company defined under the Companies Act 2009;

"information" includes information (whether in its original form or otherwise) that is in the form of a document, a signature, a seal, data, text, images, sound, or speech;

"liquidator" means a liquidator appointed under Division 2 of Part 2:

"major transaction" has the meaning set out in section 49(2) of the Companies Act 2009;

"Minister" means the Minister responsible for the administration of this Act:

"mortgagee" includes a person who derives title under the original mortgagee;

"onerous property" means:

- (a) an unprofitable contract; or
- (b) property of the company that is unsaleable, or not readily saleable, or which may give rise to a liability to pay money or perform an onerous act

"overseas company" means a corporation that is incorporated outside Solomon Islands:

"preferential claim" means a claim referred to in Part 3 of Schedule 8 (except clause 15 of that Schedule);

"prescribed form" means a form prescribed by regulations or, if no

form is prescribed by regulations, a form approved by the Registrar;

"private company" means a company that is registered as a private company on the Solomon Islands register;

"property" includes:

- (a) real and personal property; and
- (b) an estate or interest in real or personal property; and
- (c) a debt; and
- (d) any thing in action; and
- (e) any other rights, interests, and claims of any kind in relation to property;

"property in receivership" means property in respect of which a receiver is appointed;

"proponent" means a person who proposed a compromise in accordance with Division 1 of Part 2;

"public company" means a company that is registered as a public company on the Solomon Islands register;

"receiver" means a receiver, or a manager, or a receiver and manager in respect of any property appointed:

- (a) by or under any deed or agreement; or
- (b) by the Court in the exercise of a power conferred on the Court or in the exercise of its inherent jurisdiction,
 - whether or not the person appointed is empowered to sell any of the property in receivership; but does not include:
- (c) a mortgagee who, whether personally or through an agent, exercises a power:
 - (i) to receive income from mortgaged property; or

- (ii) to enter into possession or assume control of mortgaged property; or
- (iii) to sell or otherwise alienate mortgaged property; or
- (d) an agent of any such mortgagee;

"registered document" means a document:

- (a) that forms part of the register referred to in section 188(1) of the *Companies Act 2009*;
- (b) details of which have been entered in any device or facility referred to in section 188(2) of the Companies Act 2009;
- "registered office", in relation to a company, has the meaning set out in section 17 of the Companies Act 2009;
- "Registrar" means the Registrar of Companies appointed under the Companies Act 2009;
- "related company" means a related company as defined under the Companies Act 2009;
- "shareholder" means a person whose name is entered on the share register of a company as the holder of 1 or more shares in the company;
- "Solomon Islands register" means the register kept and maintained by the Registrar under the Companies Act 2009;
- "solvency test" means the solvency test referred to in the Companies Act 2009;

"special resolution" means a resolution:

- (a) approved in accordance with section 51 of the *Companies Act 2009*; or
- (b) approved at a meeting of shareholders called to consider that resolution on not less than 10 working days' notice:
 - (i) by a majority of 75% (or such higher majority as may

be specified in the rules) of the votes of shareholders entitled to vote and voting on the question; and

(ii) in accordance with any additional requirements specified in the rules in respect of such resolutions;

"subsidiary" mean a subsidiary referred to in the Companies Act 2009;

"telecommunications services":

- (a) means the conveyance by electromagnetic means from 1 device to another of any encrypted or non-encrypted sign, signal, impulse, writing, image, sound, instruction, information, or intelligence of any nature, whether for the information of any person using the device or not; but
- (b) does not include any conveyance that constitutes broadcasting;

"working day" means a day of the week other than:

- (a) Saturday and Sunday; or
- (b) a day that is defined as, or declared to be, a public holiday under any other written law;

"writing" includes representing or reproducing words, figures, or symbols in a visible and tangible form by any means and in any medium or in a visible form in any medium by electronic means that enables them to be stored in permanent form and be retrieved and read.

2 Public notice

If, under this Act, public notice must be given, that notice must be given by publishing it in at least 1 issue of a newspaper circulating widely in Solomon Islands and

(a) in the case of a matter affecting a grantor that is not an overseas company, at least 2 issues of any newspaper circulating in the area in which is situated:

- (i) the grantor's place of business; or
- (ii) if the grantor has more than 1 place of business, the grantor's principal place of business; or
- (iii) if the grantor has no place of business or neither its place of business nor its principal place of business is known, the grantor's registered office in the case of a body corporate, or the residence of the grantor in the case of an individual; or
- (b) in the case of a matter affecting a grantor that is an overseas company, at least 2 issues of any newspaper circulating in the area in which is situated:
 - (i) the place of business in Solomon Islands of the grantor; or
 - (ii) if the grantor has more than 1 place of business in Solomon Islands, the principal place of business in Solomon Islands of the grantor.

SCHEDULE 2

(Section 13)

MEETINGS OF CREDITORS

Part 1—General Provisions

1 Procedure generally

Except as provided in this Schedule and in any regulations made under this Act, a meeting of creditors may regulate its own procedure.

2 Effect of irregularity or defect

- (1) An irregularity or defect in the proceedings at a meeting of creditors does not invalidate anything done by a meeting of creditors, unless the Court orders otherwise.
- (2) The Court may, on the application of a liquidator, or a creditor of the company, make an order under subclause (1) if it is

52

satisfied that substantial injustice would be caused if the order were not made.

Part 2—Methods of Holding Meetings

3 Methods of holding meetings

A meeting of creditors may be held:

- (a) by assembling together those creditors entitled to take part and who choose to attend at the place, date, and time appointed for the meeting by the person convening the meeting as being in his or her opinion the most convenient place, date, and time for the majority of creditors; or
- (b) by means of audio, or audio and visual, communication by which all creditors participating may simultaneously hear each other throughout the meeting; or
- (c) by conducting a postal ballot, in accordance with clauses 19 to 24, of those creditors entitled to take part.

Part 3—Notice of Meeting

4 Notice of meeting

At least 5 working days before a creditors meeting, written notice must be sent to every creditor entitled to attend the meeting of:

- (a) the time and place of every meeting to be held under clause 3(a); or
- (b) the time and method of communication for every meeting to be held under clause 3(b); or
- (c) the time and address for the return of voting papers for every meeting to be held under clause 3(a), (b), or (c).

5 Contents of notice

The notice must:

(a) state the nature of the business to be transacted at the

- meeting in sufficient detail to enable a creditor to form a reasoned judgment in relation to it; and
- (b) set out the text of any resolution to be submitted to the meeting; and
- (c) include a voting paper in respect of each such resolution and voting and mailing instructions; and
- (d) state that if a creditor votes by casting a postal vote in respect of a resolution that is to be submitted to the meeting and a different resolution is submitted to the meeting:
 - (i) the creditor's postal vote is invalid in respect of that different resolution; but
 - (ii) the creditor may vote, in respect of that different resolution, either by being present in person or by proxy.

6 Effect of irregularity, etc. in notice

An irregularity in or a failure to receive a notice of a meeting of creditors does not invalidate anything done by a meeting of creditors if:

- (a) the irregularity or failure is not material; or
- (b) all the creditors entitled to attend and vote at the meeting attend the meeting without protest as to the irregularity or failure; or
- (c) all such creditors agree to waive the irregularity or failure.

Part 4—Meeting

7 Adjournment of meeting

- (1) If the meeting of creditors agrees, the chairperson may adjourn the meeting from time to time and from place to place.
- (2) An adjourned meeting must be held in the same place unless

54

another place is specified in the resolution for the adjournment.

(3) If a meeting of creditors under clause 3(a) or (b) is adjourned for less than 30 days, it is not necessary to give notice of the time and place of the adjourned meeting other than by announcement at the meeting that is adjourned.

8 Chairperson

- (1) If a liquidator has been appointed and is present, or if the liquidator has appointed a nominee and the nominee is present, he or she must act as the chairperson of a meeting held in accordance with clause 3(a) or (b).
- (2) At any meeting of creditors, if the liquidator or any nominee of the liquidator, as is applicable, is not present, or if there is no liquidator holding office for the time being, the creditors participating must choose 1 of their number to act as the chairperson of the meeting.
- (3) The person convening a meeting under clause 3(c) must do everything necessary that would otherwise be done by the person chairing a meeting.

9 Quorum

- (1) A guorum for a meeting of creditors is present if:
 - (a) 3 creditors who are entitled to vote or their proxies are present or have cast postal votes; or
 - (b) if the number of creditors entitled to vote does not exceed 3, the creditors who are entitled to vote or their proxies are present or have cast postal votes.
- (2) If a quorum is not present within 30 minutes after the time appointed for the meeting, the meeting is adjourned to the same day in the following week at the same time and place, or to such other date, time, and place as the chairperson may appoint, and if, at the adjourned meeting, a quorum is not present within 30 minutes after the time appointed for the meeting, the creditors present or their proxies are a quorum.

10 Corporations may act by representatives

A body corporate that is a creditor may appoint a representative to attend a meeting of creditors on its behalf.

11 Keeping of record of attendance and minutes

- (1) The chairperson of a meeting of creditors, or in the case of a meeting held under clause 3(c), the person convening the meeting must:
 - (a) ensure that an accurate record is kept of all creditors present or represented at the meeting, including:
 - the name of each creditor present or represented;
 and
 - (ii) whether the creditor has made a claim, and the amount of the claim; and
 - (iii) whether the creditor has filed a proxy or is present in person; and
 - (iv) the total number of creditors present or represented; and
 - (b) ensure that minutes are kept of all proceedings.
- (2) Records of attendance or minutes that have been signed correct by the chairperson or the person convening the meeting are *prima facie* evidence of the details recorded and proceedings of the meeting.

Part 5—Proxies

12 Proxies

- (1) A creditor may exercise the right to vote either by being present in person or by proxy.
- (2) A proxy for a creditor is entitled to attend and be heard at a meeting of creditors as if the proxy were the creditor.

- (3) A proxy must be appointed by notice in writing signed by the creditor and the notice must state whether the appointment is for a particular meeting or a specified term not exceeding 12 months.
- (4) No proxy is effective in relation to a meeting unless a copy of the notice of appointment is delivered to the liquidator, or if no liquidator is acting in respect of a company in liquidation, to the person by whom the notice convening the meeting was given, not less than 2 working days before the start of the meeting.

13 Liquidator may act as proxy

- A creditor may appoint any person, including the liquidator or, if there is no liquidator, the chairperson of a meeting, to act as his or her proxy.
- (2) Subject to a direction of a meeting of creditors, a liquidator must not solicit for proxies.
- (3) Without limiting the orders that the Court may make, if a liquidator has not complied with subclause (2), the Court may:
 - (a) order that the liquidator is not entitled to his or her remuneration; or
 - (b) make an order removing the liquidator from office; or
 - (c) make an order declaring any transaction entered into by the liquidator to be void or overturning any vote, and granting such consequential relief as the Court thinks fit.

14 Irregularity in notice of proxy

If an irregularity that is not material is contained in the notice of proxy, the liquidator or chairperson of a meeting, as the case may be, may accept the proxy as being valid for voting purposes, if he or she is satisfied that the proxy holder represents the creditor.

15 Limits on holder of proxy

(1) Subject to subclause (2), no person acting under a proxy may vote in favour of or against any resolution that would place that

- person, either directly or indirectly, in a position to receive any benefit out of the assets of the company otherwise than as a creditor rateably with the other creditors of the company.
- (2) Any person who holds a proxy to vote for the appointment of a liquidator may use the proxy to vote in favour of the appointment of himself or herself as liquidator if it is not inconsistent with the terms of the proxy to do so.
- (3) If a liquidator who holds a proxy cannot attend a meeting of creditors called under this Act, he or she may, in writing, nominate his or her partner (if the liquidator or is a member of a partnership) or some person in his or her employment, to use the proxy on his or her behalf and in such manner as he or she may direct.
- (4) Nothing in subclause (3) authorises the person nominated to vote in a manner that would be in contravention of subclauses(1) and (2) if the liquidator had acted under the proxy personally.

Part 6—Voting

16 Entitlement to vote, etc. determined by chairperson

- (1) For the purpose of determining whether a person is allowed to vote at a meeting and the value of the person's claim for voting purposes, the chairperson has the power to determine, for the purpose of the meeting:
 - (a) that the person is a creditor of the company; and
 - (b) the value of a creditor's claim against the company.
- (2) If the chairperson is uncertain as to whether a person is a creditor of the company or as to the value of the person's claim against the company, the chairperson must allow the person to vote subject to the vote being subsequently declared invalid in whole or in part by the chairperson.
- (3) A creditor who is not entitled to vote, may, with the leave of the liquidator, attend and speak at a meeting of creditors.

(4) A creditor chairing the meeting does not have a casting vote.

17 Voting by secured creditors

In the case of a meeting of creditors held under Division 3 of Part 2 of this Act (liquidations):

- (a) a secured creditor is entitled to vote:
 - (i) for the whole debt if he or she surrenders the security interest to the liquidator for the general benefit of creditors; or
 - (ii) in respect of the balance of the debt if he or she values the security interest and claims as an unsecured creditor for the balance due; or
 - (iii) in respect of the balance of the debt if he or she realises property subject to a security interest and claims as an unsecured creditor for any balance due after deducting the net amount realised;
- subject to this Act, if a secured creditor votes in respect of the creditor's whole debt, the creditor is taken to have surrendered his or her security interest;
- (c) a creditor who is not entitled to vote may, with the leave of the liquidator, attend and speak at the meeting.

18 When resolution adopted

At any meeting of creditors or a class of creditors, a resolution is adopted if a majority in number and value of the creditors or the class of creditors voting in person or by proxy or by postal vote, vote in favour of the resolution.

Part 7—Postal voting

19 Who may cast postal vote

A creditor entitled to vote at a meeting of creditors held in accordance with clause 3(a), (b), or (c) may exercise the right to vote by casting a postal vote in relation to a matter to be decided at that meeting.

20 Postal vote cast in respect of different resolution

If a creditor votes by casting a postal vote in respect of a resolution that is to be submitted to the meeting and a different resolution is submitted to the meeting:

- (a) the creditor's postal vote is invalid in respect of that different resolution; but
- (b) the creditor may vote, in respect of that different resolution, either by being present in person or by proxy.

21 Person authorised to receive and count postal votes

- (1) The notice of meeting must state the name of the person authorised to receive and count postal votes in relation to that meeting.
- (2) If no person has been authorised to receive and count postal votes in relation to a meeting, or if no person is named as being so authorised in the notice of the meeting, the liquidator is deemed to be so authorised.

22 How to cast postal vote

A creditor may cast a postal vote on all or any of the matters to be voted on at the meeting by sending a marked voting paper to a person authorised to receive and count postal votes in relation to that meeting, so as to reach that person not less than 2 working days before the start of the meeting or, if the meeting is held under clause 3(c), not later than the date named for the return of the voting paper.

23 Duty of person authorised to receive and count postal votes

- (1) It is the duty of a person authorised to receive and count postal votes in relation to a meeting:
 - (a) to collect together all postal votes received by him or her;
 and
 - (b) in relation to each resolution to be voted on:

- to count the number of creditors or creditors belonging to a class of creditors, as the case may be, voting in favour of the resolution and determine the total amount of the debts owed by the company to those creditors; and
- (ii) to count the number of creditors or creditors belonging to a class of creditors, as the case may be, voting against the resolution and determine the total amount of the debts owed by the company to those creditors: and
- (c) to sign a certificate:
 - (i) that he or she has carried out the duties set out in paragraphs (a) and (b); and
 - (ii) stating the results of the counts and determinations required by paragraph (b); and
- (d) to ensure that the certificate required by paragraph (c) is presented to the person chairing or convening the meeting.
- (2) A certificate given under subclause (1) in relation to the postal votes cast in respect of a meeting of creditors must be annexed to the minutes of the meeting.

24 Duty of chairperson

If a vote is taken at a meeting held under clause 3(a) or (b) on a resolution on which postal votes have been cast, the person chairing the meeting must include the results of voting by all creditors who have sent in a voting paper duly marked as for or against the resolution.

SCHEDULE 3

(Section 13)

POWERS, FUNCTIONS, AND LIABILITIES OF LIQUIDATORS

Part 1—Preliminary Provisions

1 Power to appoint 2 or more liquidators

For the purposes of this Act, the power to appoint a liquidator of a company includes the power to appoint 2 or more persons as liquidators of a company.

2 Liquidators to act jointly

If 2 or more persons are appointed as liquidators of a company, those persons must act jointly unless the special resolution of shareholders, the resolution of the directors of the company, or the order of the Court appointing the liquidators states that the liquidators may exercise their powers individually.

3 When liquidator not required to act

Despite anything in this Act, except if the security interest is surrendered or taken to be surrendered or redeemed, a liquidator may, but is not required to, carry out any duty or exercise any power in relation to property that is subject to a security interest.

Part 2—Powers of Liquidators

4 Liquidator controls company's assets

With effect from the commencement of the liquidation of a company, the liquidator has custody and control of the company's assets.

5 General powers

A liquidator has the powers:

- (a) necessary to carry out the functions and duties of a liquidator under this Act; and
- (b) conferred on a liquidator by this Act.

6 Specific powers

Without limiting clause 5, a liquidator of a company has power to:

- (a) commence, continue, discontinue, and defend legal proceedings;
- (b) carry on the business of the company, to the extent necessary for the liquidation;
- (c) appoint a legal practitioner;
- (d) pay any class of creditors in full;
- (e) make a compromise or an arrangement with creditors or persons claiming to be creditors or who have or allege the existence of a claim against the company, whether present or future, actual or contingent, or ascertained or not:
- (f) compromise calls and liabilities for calls, debts, and liabilities capable of resulting in debts, and claims, present or future, actual or contingent, ascertained or not, subsisting or supposed to subsist between the company and any person and all questions relating to, or affecting the assets or the liquidation of, the company, on such terms as may be agreed, and take security for the discharge of any such call, debt, liability, or claim, and give a complete discharge;
- (g) sell or otherwise dispose of the property of the company;
- (h) act in the name and on behalf of the company and enter into deeds, contracts, and arrangements in the name and on behalf of the company;
- (i) prove, rank, and claim in the bankruptcy or insolvency of a debtor or shareholder for any balance against that person's estate, and to receive dividends in the bankruptcy or insolvency, as a separate debt due from the bankrupt or insolvent and rateably with the other separate creditors;
- (j) draw, accept, make, and endorse a bill of exchange or

promissory note in the name and on behalf of the company, with the same effect as if the bill or note had been drawn, accepted, made, or endorsed by or on behalf of the company in the course of its business;

- (k) borrow money on the security of the company's assets;
- (I) take out, in his or her name as liquidator, letters of administration to a deceased shareholder, and to do in that name any other act necessary for obtaining payment of money due from a debtor or shareholder or his or her estate, that cannot be conveniently done in the name of the company. In all such cases, the money due is, for the purpose of enabling the liquidator to take out the letters of administration or recover the money, deemed to be due to the liquidator;
- (m) call a meeting of creditors or shareholders for:
 - the purpose of informing creditors or shareholders of progress in the liquidation;
 - (ii) the purpose of ascertaining the views of creditors or shareholders on any matter arising in the liquidation;
 - (iii) such other purpose connected with the liquidation as the liquidator thinks fit; and
- (n) appoint an agent to do anything that the liquidator is unable to do.

7 Liquidator may enforce liability of shareholders

A liquidator may enforce the liability of the shareholder or former shareholder in respect of any shares issued to the shareholder or former shareholder.

8 Liquidator may disclaim onerous property

(1) Subject to clause 9, a liquidator may disclaim onerous property even though the liquidator has taken possession of it, tried to sell it, or otherwise exercised rights of ownership in relation to

(2) A disclaimer:

- (a) brings to an end, on and from the date of the disclaimer, the rights, interests, and liabilities of the company in relation to the property disclaimed;
- (b) does not, except so far as necessary to release the company from a liability, affect the rights or liabilities of any other person.
- (3) A liquidator who disclaims onerous property must, within 10 working days of the disclaimer, give notice in writing of the disclaimer to any person whose rights are, to the knowledge of the liquidator, affected by the disclaimer.
- (4) A person suffering loss or damage as a result of a disclaimer under this clause may:
 - (a) claim as a creditor of the company for the amount of the loss or damage, taking account of the effect of an order made by the Court under paragraph (b);
 - (b) apply to the Court for an order that the disclaimed property be delivered to, or vested in, that person.
- (5) The Court may make an order under subclause (4)(b) if it is satisfied that it is just that the property should be vested in the applicant.

9 Liquidator may be required to elect whether to disclaim onerous property

A liquidator is not entitled to disclaim onerous property if:

(a) a person whose rights would be affected by the disclaimer of onerous property gives the liquidator notice in writing requiring the liquidator to elect whether to disclaim the onerous property before the close of a date specified in the notice, which must be at least 20 working days after the date on which the notice is received by the liquidator; and

(b) the liquidator does not disclaim the onerous property before the close of that date.

Part 3—Duties of Liquidators

10 Principal duties of liquidator

The principal duties of a liquidator of a company are, in a reasonable and efficient manner:

- (a) to take possession of, protect, realise, and distribute the assets, or the proceeds of the realisation of the assets, of the company to its creditors in accordance with this Act; and
- (b) if there are surplus assets remaining, to distribute them, or the proceeds of the realisation of the surplus assets, in accordance with section 51.

11 Restriction on purchase of company's assets by liquidator

- (1) Subject to the leave of the Court, a liquidator must not, either directly or indirectly, become a purchaser of any part of the company's assets.
- (2) The Court may set aside any purchase made contrary to this clause, and grant any consequential relief that it thinks fit.
- (3) The Court may give its leave on any conditions that it thinks fit.

12 Restriction on purchase of goods or services from persons connected with liquidator

- (1) Subject to the leave of the Court, a liquidator must not purchase goods or services for the purposes of the liquidation from any person whose connection with him or her would result in the liquidator directly or indirectly obtaining any benefit arising out of the transaction.
- (2) The Court may disallow or recover any benefit made contrary to this clause.

(3) The Court may give its leave on any conditions that it thinks fit.

13 Deposit of company funds

A liquidator must deposit the funds of a company under his or her administration in a bank account to the credit of the company or in a trust account at a bank on trust for the benefit of the company.

14 Investment of funds

- (1) Despite clause 13, in any liquidation all or any part of the balance standing to the credit of the company in any bank account or trust account kept by the liquidator, and not required for the time being to meet claims made against the company, may be invested in any bank or in any Government securities or, if authorised by the Court, any other securities.
- (2) All dividends, interest, and other profits from investments must immediately on being received be paid into the bank account or trust account kept by the liquidator under clause 13.

15 Duties in relation to accounts

- (1) Subject to subclause (2), the liquidator of a company must:
 - (a) keep accounts and records of the liquidation and permit those accounts and records, and the accounts and records in the company, to be inspected by:
 - (i) any appointed liquidation committee, unless the liquidator believes on reasonable grounds that inspection would be prejudicial to the liquidation; and
 - (ii) if the Court so orders, a creditor or shareholder; and
 - (b) retain the accounts and records of the liquidation and of the company for not less than 1 year after completion of the liquidation; and
 - (c) if a liquidator carries on the business of the company, keep accounting records for the carrying on of the business of the company that comply with section 124 of

the *Companies Act 2009* to the extent that that section is applicable.

- (2) The Registrar may, whether before or after the completion of the liquidation:
 - (a) authorise the disposal of any accounts and records; and
 - (b) require any accounts or records to be retained for longer than 1 year after the completion of the liquidation.

16 Meaning of failure to comply

In clauses 17 to 19, failure to comply means a failure of a liquidator to comply with a relevant duty arising:

- (a) under this Act or other written laws or other law; or
- (b) under any order or direction of the Court other than an order to comply made under clause 17.

17 Failure to comply

If the Court is satisfied that there is, or has been, a failure to comply, the Court may:

- (a) relieve the liquidator of the duty to comply wholly or in part; or
- (b) without prejudice to any other remedy that may be available in relation to a breach of duty by the liquidator, order the liquidator to comply to the extent specified in the order.

18 Consequences of non-compliance with Court order

The Court may, in relation to a person who fails to comply with an order made under clause 17, or is or becomes disqualified to become or remain a liquidator:

- (a) remove the liquidator from office; or
- (b) order that the person may be appointed and act, or may

68

continue to act, as liquidator, despite being disqualified to act as liquidator.

19 Prohibition order

- (1) The Court must make, in relation to a person, a prohibition order for a period not exceeding 5 years if it is shown to the satisfaction of the Court that the person is unfit to act as liquidator by reason of:
 - (a) persistent failures to comply; or
 - (b) the seriousness of a failure to comply.
- (2) A person to whom a prohibition order applies must not:
 - (a) act as a liquidator in a current or other liquidation; or
 - (b) act as a receiver in a current or other receivership.
- (3) The following is, in the absence of special reasons to the contrary, evidence of persistent failures to comply for the purposes of this clause:
 - (a) evidence that on 2 or more occasions within the preceding 5 years, the Court has made an order to comply under this clause in respect of the same person;
 - (b) evidence that on 2 or more occasions within the preceding 5 years, an application for an order to comply under this clause has been made in respect of the same person and that in each case the person has complied after the making of the application and before the hearing.
- (4) A copy of every prohibition order must, within 10 working days of the order being made, be delivered by the applicant to the Registrar, who must keep it on a file indexed by reference to the name of the liquidator concerned.

20 Who may apply for orders under clauses 17 to 19

(1) An application for an order under this Part may be made by:

- (a) a liquidator;
- (b) a person seeking appointment as a liquidator;
- (c) a liquidation committee;
- (d) a creditor, shareholder or a director of the company in liquidation;
- (e) a receiver appointed in relation to property of the company in liquidation;
- (f) the Registrar.
- (2) No application may be made to the Court by a person other than a liquidator in relation to a failure to comply unless notice of the failure to comply has been served on the liquidator not less than 5 working days before the date of the application and, as at the date of the application, there is a continuing failure to comply.

21 Court orders under clauses 17 to 19: general

In making an order under this clause, the Court may:

- (a) make an order extending the time for compliance; or
- (b) impose terms or conditions; or
- (c) make an ancillary order.

Part 4—Court Supervision of Liquidations

22 Court orders

On the application of the liquidator, a liquidation committee, or, with the leave of the Court, a creditor, shareholder, or director of a company in liquidation, the Court may:

- (a) give directions in relation to any matter arising in connection with the liquidation; or
- (b) confirm, reverse, or modify an act or decision of the liquidator; or

- (c) order an audit of the accounts of the liquidation; or
- (d) order the liquidator to produce the accounts and records of the liquidation for audit and to provide the auditor with the information concerning the conduct of the liquidation that the auditor requests; or
- in respect of any period, review or fix the remuneration of the liquidator at a level that is reasonable in the circumstances; or
- (f) to the extent that an amount retained by the liquidator as remuneration is found by the Court to be unreasonable in the circumstances, order the liquidator to refund the amount; or
- (g) declare whether or not the liquidator was validly appointed or validly assumed custody or control of property; or
- (h) make an order concerning the retention or the disposition of the accounts and records of the liquidation or of the company.

23 Court orders are additional to other Court powers

The powers given by clause 22:

- (a) are in addition to any other powers the Court may exercise in its jurisdiction relating to liquidators under this Act; and
- (b) may be exercised:
 - in relation to a matter occurring either before or after the commencement of the liquidation or the removal of the company from the Solomon Islands register; and
 - (ii) whether or not the liquidator has ceased to act as liquidator when the application or the order is made.

24 Defence to act in accordance with Court direction

- (1) Subject to subclause (2), a liquidator is entitled to rely on having so acted as a defence to a claim in relation to anything done or not done in accordance with the direction if the liquidator has:
 - (a) obtained a direction of the Court with respect to a matter connected with the exercise of the powers or functions of liquidator; and
 - (b) acted in accordance with the direction.
- (2) The Court may, on the application of any person, order that, by reason of the circumstances in which a direction by the Court was obtained, the liquidator does not have the protection given by subclause (1).

SCHEDULE 4

(Section 13)

OFFICE OF LIQUIDATOR

Part 1—Restrictions on Appointment of Liquidators

1 Who may not be appointed or act as liquidator

None of the following may be appointed or act as a liquidator of a company:

- (a) a corporation;
- (b) a person who is under 21 years of age;
- (c) a creditor of the company;
- (d) a person who has, within the 2 years immediately before the beginning of the liquidation, been a shareholder, employee, director, auditor, or receiver of the company or of a related company;
- (e) an undischarged bankrupt;
- (f) a person in respect of whom an order of custody is in force under section 18 or 41 of the *Mental Treatment Act* (Cap

72

103);

(g) a person who is prohibited from being a director or promoter, or being concerned or taking part in the management, of a company under this Act.

2 Validity of acts of liquidators

- (1) The acts of a person as a liquidator are valid even though that person may not be qualified to act as a liquidator.
- (2) No defect or irregularity in the appointment of a liquidator invalidates any act done by him or her in good faith.

3 Person must consent to being appointed liquidator

A person must not be appointed as liquidator of a company unless:

- (a) the person has consented in writing to the appointment; and
- (b) as at the time of the appointment, the person has not withdrawn the consent.

4 Court may declare whether liquidator validly appointed

- (1) If there is doubt on a specific ground about whether the appointment of a person as liquidator of a company is valid, the person, the company, or any of the company's creditors may apply to the Court for an order under subclause (2).
- (2) The Court may, on application, make an order declaring whether or not the appointment was valid on the ground specified in the application or on some other ground.

Part 2—Vacancy in Office of Liquidator

5 Vacancy in office of liquidator

The office of liquidator becomes vacant if the person holding office resigns, dies, or is or becomes disqualified to act as liquidator.

6 How liquidator may resign

A person may resign from the office of liquidator by appointing another person as his or her successor and sending or delivering notice in writing of the appointment of his or her successor to the Registrar for registration.

7 Court may review appointment of successor

The Court may, on the application of the company, or a shareholder or other entitled person, or a director or creditor of the company, review the appointment of a successor to a liquidator and may appoint any person who could be appointed as liquidator under section 16, 17, or 18, as the case may be, to be the liquidator of the company.

8 Vacancy not caused by resignation

If, for any reason other than resignation, a vacancy occurs in the office of liquidator, written notice of the vacancy must immediately be sent or delivered to the Registrar by the person vacating office or, if that person is unable to act, by his or her personal representative.

9 Appointment of liquidator until successor appointed

If, as the result of the vacation of office by a liquidator, no person is acting as liquidator, the Registrar may appoint a person to act as liquidator until a successor is appointed under this clause.

10 Appointment of successor by Court

If a vacancy occurs in the office of the liquidator, or a liquidator has been appointed under clause 9, as the case may be, the Court may, on the application of the company, or a shareholder or other entitled person, or a director or creditor of the company, or the Registrar, appoint any person who could be appointed as liquidator under section 16, 17, or 18, as the case may be, to be the liquidator of the company.

11 Notice of appointment given by successor

A liquidator appointed under clause 10 must, within 10 working days of being appointed or being notified of his or her appointment, deliver a notice of his or her appointment to the Registrar for registration.

12 Vacating liquidator's successor to be helped

- (1) A person vacating the office of liquidator must, if practicable, provide such information and give such assistance to that person's successor as he or she reasonably requires in taking over the duties of liquidator.
- (2) A person vacating the office of liquidator must immediately, or within any reasonable time that may be specified by that person's successor, deliver to his or her successor the following things that are in his or her possession or under his or her control:
 - (a) any records or documents of the company;
 - (b) other property of the company;
 - (c) all claims;
 - (d) accounts and records of the liquidation.

13 Liquidator ceases to hold office on completion of liquidation

- (1) A liquidator ceases to hold office on the completion of the liquidation.
- (2) Subclause (1) does not limit Part 3 or 4 of Schedule 3.

Part 3—Liquidators' Remuneration

14 Remuneration of liquidators

- (1) Subject to clause 22(f) of Schedule 3, a liquidator appointed under section 16 or 17 is entitled to charge reasonable remuneration for carrying out his or her duties and exercising his or her powers as liquidator.
- (2) Unless the Court otherwise orders, a liquidator appointed under section 18 must charge remuneration either:
 - (a) of an amount equal to the amount fixed under regulations made under this Act; or

(b) at, or in accordance with, such rate or rates as may be prescribed under regulations made under this Act.

15 Expenses and remuneration payable out of assets of company

The expenses and remuneration of the liquidator are payable out of the assets of the company.

SCHEDULE 5

(Section 13)

EFFECT OF LIQUIDATION

Part 1—Preliminary Provision

1 Company's rules not to be altered

With effect from the commencement of the liquidation of a company, the rules of the company cannot be altered.

Part 2—Effect on Company's Officers and Shareholders, etc.

2 Functions and powers of company's directors suspended

With effect from the commencement of the liquidation of a company, the directors remain in office but cease to have powers, functions, or duties other than those required or permitted to be exercised by this Act.

3 Effect on company's shareholders

With effect from the commencement of the liquidation of a company:

- (a) unless the Court orders otherwise, a share in the company must not be transferred; or
- (b) an alteration must not be made to the rights or liabilities of a shareholder or former shareholder of the company;
 or
- (c) a shareholder must not exercise a power under the rules

76

of the company or this Act, except for the purposes of this Act.

Part 3—Effect on Proceedings

4 Legal proceedings not to be commenced or continued

With effect from the commencement of the liquidation of a company, a person must not, unless the liquidator agrees or the Court orders otherwise, commence or continue legal proceedings against the company or in relation to its property.

5 Effect on proceedings commenced before commencement of liquidation

- (1) At any time after the making of an application to the Court to appoint a liquidator of a company and before a liquidator is appointed, the company or any creditor or shareholder of the company may:
 - (a) in the case of any application or proceeding against the company that is pending in the Court or Court of Appeal, apply to the Court or Court of Appeal, as the case may be, for a stay of the application or proceeding; or
 - (b) in the case of any other application or proceeding pending against the company in any other court or tribunal, apply to the Court to restrain the application or proceeding.
- (2) The Court or Court of Appeal, as the case may be, may stay or restrain the application or proceedings on any terms that it thinks fit.

6 No enforcement of rights over company's property

- (1) With effect from the commencement of the liquidation of a company, a person must not, unless the liquidator agrees or the Court orders otherwise, exercise or enforce, or continue to exercise or enforce, a right or remedy over or against property of the company.
- (2) Nothing in subclause (1) or clause 4 affects the right of a

secured creditor to take possession of, realise or otherwise deal with, property of the company over which that creditor has a security interest.

7 Restriction on rights of creditors to complete execution, distraint, or attachment

- (1) Subject to subclauses (2) and (3), a creditor is not entitled to retain the benefit of any execution process, distress, or attachment over or against the property of a company unless the execution process, distress, or attachment is completed before:
 - (a) the passing of a special resolution appointing a liquidator of the company, or the date on which the creditor had notice of the calling of a meeting at which such a resolution was proposed, whichever occurs first; or
 - (b) the passing of a resolution by the directors of a company appointing a liquidator of the company, or the date on which the creditor had notice of the calling of a meeting at which such a resolution was proposed, whichever occurs first; or
 - (c) the making of an application to the Court to appoint a liquidator of the company.

(2) Despite subclause (1):

- (a) a person who, in good faith, purchases property of a company from a court officer charged with an execution process acquires a good title as against the liquidator of the company;
- (b) a person who, in good faith, purchases property of a company on which distress has been levied acquires a good title as against the liquidator of the company.
- (3) The Court may set aside the application of subclause (1) to the extent and on any conditions that the Court thinks fit.
- (4) For the purposes of this clause:

- (a) an execution or distraint against personal property is completed by seizure and sale; or
- (b) an attachment of a debt is completed by receipt of the debt; or
- (c) an execution against land is completed by sale, and, in the case of an equitable interest, by the appointment of a receiver.
- (5) Nothing in this section limits or affects Part 2 of Schedule 7.

8 Duties of court officer in execution process

- (1) A court officer must, on being required by a liquidator of a company to do so, deliver or transfer the company's property and any money received in satisfaction or partial satisfaction of an execution or paid to avoid a sale of the property, as the case may be, to the liquidator if:
 - (a) the property has been taken in an execution process; and
 - (b) before completion of the execution process, the court officer charged with the execution process receives notice that the liquidator of the company has been appointed.
- (2) The costs of the execution process are a first security interest on any property or money delivered or transferred to the liquidator under subclause (1) and the liquidator may sell all or some of the property to satisfy that security interest.
- (3) The court officer must retain the proceeds of sale or the money paid for 10 working days if:
 - (a) property of a company is sold in an execution process in respect of a judgment for a sum exceeding \$500; or
 - (b) money is paid to the court officer charged with the execution process to avoid a sale of the property.
- (4) The court officer must deduct from the amount the costs of the execution process and pay the balance to the liquidator if:

- (a) within the period of 10 working days, the court officer has notice of:
 - (i) the calling of a meeting at which a special resolution is proposed to appoint a liquidator; or
 - (ii) the calling of a meeting of the directors at which a resolution is proposed to appoint a liquidator or of a meeting of the directors at which the appointment of a liquidator is to be considered; or
 - (iii) the making of an application to the Court to appoint a liquidator; and
- (b) The company is put into liquidation.
- (5) A liquidator to whom money is paid under subclause (4) is entitled to retain it as against the execution creditor.
- (6) The Court may set aside the application of this section to the extent, and on any conditions, that it thinks fit.

Part 4—Effect on Certain Conduct

9 Certain conduct prohibited

- (1) If a company is in liquidation, or an application has been made to the Court for an order that a company be put into liquidation, as the case may be, no person may:
 - (a) leave Solomon Islands with the intention of:
 - (i) avoiding payment of money due to the company; or
 - (ii) avoiding examination in relation to the affairs of the company; or
 - (iii) avoiding compliance with an order of the Court or some other obligation under this Part in relation to the affairs of the company; or
 - (b) conceal or remove property of the company with the intention of preventing or delaying the liquidator taking

custody or control of it; or

- (c) destroy, conceal, or remove records or other documents of the company.
- (2) A person who does not comply with subclause (1) commits an offence and is liable on conviction to a fine not exceeding 12,500 penalty units or to imprisonment for a term not exceeding 2 years, or both.

SCHEDULE 6

(Sections 13 and 45)

LIQUIDATION COMMITTEES

Part 1—Liquidation Committee

1 Appointment of members

- (1) The members of a liquidation committee chosen by a meeting of creditors or of shareholders take office immediately.
- (2) The liquidator must refer the matter to the Court, and the Court may make any decision that it thinks fit, if there is a difference between the decisions of meetings of creditors and meetings of shareholders on:
 - (a) the guestion of appointing a liquidation committee; or
 - (b) the membership of a liquidation committee.

2 Membership

A liquidation committee must consist of not fewer than 3 persons who are:

- (a) creditors or shareholders; or
- (b) persons holding general powers of attorney from creditors or shareholders; or
- (c) authorised directors or representatives of companies that

81

are creditors or shareholders of the company in liquidation.

3 Powers

A liquidation committee has the power to:

- (a) call for reports from the liquidator on the progress of the liquidation; and
- (b) call a meeting of creditors or of shareholders; and
- (c) apply to the Court under Part 3 or 4 of Schedule 3; and
- (d) assist the liquidator, as appropriate, in the conduct of the liquidation.

4 Application of company's rules

A meeting of shareholders called under clause 3(b) must be held in accordance with the company's rules (except that the liquidator has power to give notice of a meeting of shareholders and to act as, or appoint, the chairperson of the meeting).

5 Inability to act

If, by reason of vacancies in a liquidation committee, the committee is unable to act, the liquidator must call attention to the situation in the next six-monthly report required to be prepared and sent under section 48.

6 Restriction on purchase of company's assets by liquidation committee

- (1) Subject to the leave of the Court, a member of a liquidation committee of a company must not, either directly or indirectly, become a purchaser of any part of the company's assets.
- (2) The Court may set aside any purchase made contrary to this clause, and grant any consequential relief that it thinks fit.
- (3) The Court may give its leave on any conditions that it thinks fit.

7 Members not entitled to benefit from dealings with company's assets

- (1) Subject to the leave of the Court, no member of a liquidation committee may directly or indirectly be entitled to:
 - (a) derive any benefit from any transaction arising out of the assets of the company; or
 - (b) receive out of the assets of the company any payment for services rendered by him or her in connection with the administration of the assets, or for any goods supplied by him or her to the liquidator for, or on account of, the company; or
 - (c) directly or indirectly become the purchaser of any part of the company's assets.
- (2) If the leave of the Court is sought under subclause (1) in respect of any payment for services, the leave may be given only if the services performed are of a special nature and the order must specify the nature of the services for which leave is given.

8 No remuneration

Except by the leave of the Court, no remuneration may, under any circumstances, be paid to a member of a liquidation committee for services rendered by him or her in the discharge of the duties attaching to his or her office as a member of the committee.

9 Disallowance or recovery of benefits or payments

- (1) The Court may disallow or recover any benefit or payment made contrary to clause 7 or 8.
- (2) The Court may give leave under clause 7 or 8 on any conditions that it thinks fit.

Part 2—Proceedings at Meetings

10 Frequency of meetings

(1) The committee must meet at the times as it from time to time

appoints.

(2) The liquidator or a member of the committee may also call a meeting of the committee as and when necessary.

11 Majorities

The committee may act by a majority of its members present at a meeting, but may not act unless a majority of the committee are present.

12 Resignation

A member of the committee may resign by notice in writing signed by the member and delivered to the liquidator.

13 Office becoming vacant

The office of a member of the committee becomes vacant if the member:

- (a) becomes bankrupt; or
- (b) compounds or arranges with his or her creditors; or
- (c) is absent from 3 consecutive meetings of the committee without the leave of those members who together with that member represent the creditors or shareholders, as the case may be.

14 Removal of member

- (1) A member of the committee may be removed by a resolution:
 - (a) carried at a meeting of creditors if the member represents creditors; or
 - (b) carried at a meeting of shareholders if the member represents shareholders.
- (2) At least 5 working days' notice of the resolution must be given, which states the object of the meeting.

15 Vacancy filled

A vacancy in the committee may be filled by the appointment to the committee of:

- (a) a creditor or shareholder, as the case may be; or
- (b) a person holding a general power of attorney from, or being an authorised director or representative of, a company that is a creditor or shareholder, as the case may be.

16 Committee with vacancy may act

The continuing members of the committee, if not less than 2, may act even though a vacancy exists in the committee.

SCHEDULE 7

(Section 13)

VOIDABLE TRANSACTIONS AND SECURITY INTERESTS AND RECOVERIES IN OTHER CASES

Part 1—Voidable Transactions and security interests

Division 1—Voidable Transactions

1 Definitions

In clause 2:

"restricted period" means:

- (a) the period of 6 months before the date of commencement of the liquidation together with the period commencing on that date and ending at the time at which the liquidator is appointed; and
- (b) in the case of a company that was put into liquidation by the Court, the period of 6 months before the making of the application to the Court together with the period commencing on the date of the making of that application and ending on the date on which, and at the time at which, the order of the Court was made;

- (c) the period of 6 months before the making of the application to the Court together with the period commencing on the date of the making of that application and ending on the date and at the time of the commencement of the liquidation if:
 - (i) an application was made to the Court to put a company into liquidation; and
 - (ii) after the making of the application to the Court a liquidator was appointed under section 16 or 17;

"specified period" means:

- (a) the period of 2 years before the date of commencement of the liquidation together with the period commencing on that date and ending at the time at which the liquidator is appointed; and
- (b) in the case of a company that was put into liquidation by the Court, the period of 2 years before the making of the application to the Court together with the period commencing on the date of the making of that application and ending on the date on which, and at the time at which, the order was made; and
- (c) the period of 2 years before the making of the application to the Court together with the period commencing on the date of the making of that application and ending on the date and at the time of the commencement of the liquidation if:
 - (i) an application was made to the Court to put a company into liquidation; and
 - (ii) after the making of the application to the Court a liquidator was appointed under section 16 or 17;

"transaction", in relation to a company, means:

- (a) a conveyance or transfer of property by the company; or
- (b) the giving of a security or security interest over the property of the company; or
- (c) the incurring of an obligation by the company; or

- (d) the acceptance by the company of execution under a judicial proceeding; or
- (e) the payment of money by the company, including the payment of money under a judgment or order of a court.

2 Voidable transactions

- (1) A transaction by a company is voidable on the application of the liquidator if the transaction:
 - (a) was made:
 - (i) at a time when the company was unable to pay its due debts; and
 - (ii) within the specified period; and
 - (b) enabled another person to receive more towards satisfaction of a debt than the person would otherwise have received or be likely to have received in the liquidation.
- (2) Subclause (1) does not apply if the transaction took place in the ordinary course of business.
- (3) Unless the contrary is proved, a transaction that took place within the restricted period is presumed to have been made:
 - (a) at a time when the company was unable to pay its debts;
 and
 - (b) otherwise than in the ordinary course of business.
- (4) In determining whether a transaction took place in the ordinary course of business, no account is, unless that other person knew that that was the intent or purpose of the company, to be taken of any intent or purpose on the part of a company:
 - (a) to enable another person to receive more towards satisfaction of a debt than the person would otherwise receive or be likely to receive in the liquidation; or

- (b) to reduce or cancel the liability, whether in whole or in part, of another person in respect of a debt incurred by the company; or
- (c) to contribute towards the satisfaction of the liability, whether in whole or in part, of another person in respect of a debt incurred by the company.

Division 2—Voidable security Interests

3 Definitions

In clause 4:

"restricted period" means:

- (a) the period of 6 months before the date of commencement of the liquidation together with the period commencing on that date and ending at the time at which the liquidator is appointed; and
- (b) in the case of a company that was put into liquidation by the Court, the period of 6 months before the making of the application to the Court together with the period commencing on the date of the making of the application and ending on the date on which, and at the time at which, the order of the Court was made: and
- (c) the period of 6 months before the making of the application to the Court together with the period commencing on the date of the making of that application and ending on the date and at the time of the commencement of the liquidation if:
 - (i) an application was made to the Court to put a company into liquidation; and
 - (ii) after the making of the application to the Court a liquidator was appointed under section 16 or 17;

"specified period" means:

(a) the period of 1 year before the date of commencement of

ឧឧ

- the liquidation together with the period commencing on that date and ending at the time at which the liquidator is appointed; and
- (b) in the case of a company that was put into liquidation by the Court, the period of 1 year before the making of the application to the Court together with the period commencing on the date of the making of the application and ending on the date on which, and at the time at which, the order of the Court was made; and
- (c) the period of 1 year before the making of the application to the Court together with the period commencing on the date of the making of that application and ending on the date and at the time of the commencement of the liquidation if:
 - (i) an application was made to the Court to put a company into liquidation; and
 - (ii) after the making of the application to the Court a liquidator was appointed under section 16 or 17.

4 Voidable security interests

- (1) A security interest over any property or undertaking of a company is voidable on the application of the liquidator if the security interest was given within the specified period, unless:
 - (a) the security interest secures:
 - (i) money actually advanced or paid; or
 - (ii) the actual price or value of property sold or supplied to the company; or
 - (iii) any other valuable consideration given in good faith by the grantee of the security interest at the time of, or at any time after, the giving of the security interest; or
 - (b) immediately after the security interest was given, the

company was able to pay its due debts; or

- (c) the security interest is in substitution for a security interest given before the specified period.
- (2) Unless the contrary is proved, a company giving a security interest within the restricted period is presumed to have been unable to pay its due debts immediately after giving the security interest.

5 Exception: certain kinds of substituted security interests

Clause 4(1)(c) does not apply to the extent that:

- (a) the amount secured by the substituted security interest exceeds the amount secured by the existing security interest; or
- (b) the value of the property subject to the substituted security interest at the date of the substitution exceeds the value of the property subject to the existing security interest at that date.

6 Exception: security interest that secures unpaid purchase price

Nothing in clause 4 applies to a security interest given by a company that secures the unpaid purchase price of property, whether or not the security interest is given over that property, if:

- (a) the security interest document is executed not later than 30 days after the sale of the property; or
- (b) in the case of the sale of an estate or interest in land, the security interest document is executed not later than 30 days after the final settlement of the sale.

7 Payments received by secured party

For the purposes of clauses 4(1)(a) and 6, if a security interest was given by the company within the period specified in clause 4, all payments received by the secured party entitled to the security interest after it was given are deemed to have been appropriated so

far as may be necessary:

- (a) towards repayment of money actually advanced or paid by the secured party to the company on or after the giving of the security interest; or
- (b) towards payment of the actual price or value of property sold by the secured party to the company on or after the giving of the security interest; or
- (c) towards payment of any other liability of the company to the secured party in respect of any other valuable consideration given in good faith on or after the giving of the security interest.

Division 3—Procedure for Setting Aside Voidable Transactions and Security Interests

8 Procedure

- (1) A liquidator who wishes to have a transaction that is voidable or a security interest that is voidable set aside must:
 - (a) file in the Court a notice to that effect specifying the transaction or security interest to be set aside and, in the case of a transaction, the property or value which the liquidator wishes to recover, and also the effect of subclauses (2), (3), and (4); and
 - (b) serve a copy of the notice on the other party to the transaction or the secured creditor entitled to the security interest and on any other person from whom the liquidator wishes to recover.

(2) A person:

- (a) who would be affected by the setting aside of the transaction or security interest specified in the liquidator's notice: and
- (b) who considers that the transaction or security interest is not voidable.

may file in the Court a notice objecting to the transaction or security interest being set aside, and serve a copy of that notice on the liquidator, within 20 working days after the service of the liquidator's notice.

- (3) Unless a person on whom the liquidator's notice was served has given notice under subclause (2), the transaction or security interest is set aside on the 20th working day after the date of service of the notice.
- (4) If 1 or more persons have given notice under subclause (2), the liquidator may apply to the Court for an order that the transaction or security interest be set aside. That application must be served on every person referred to in paragraph (b) of subclause (1), whether or not that person gave a notice under subclause (2).

9 Other orders

If a transaction or security interest is set aside, the Court may make 1 or more of the following orders:

- (a) an order requiring a person to pay to the liquidator, in respect of benefits received by that person as a result of the transaction or security interest, such sums as fairly represent those benefits;
- (b) an order requiring property transferred as part of the transaction to be restored to the company;
- (c) an order requiring property to be vested in the company if it represents in a person's hands the application, either of the proceeds of sale of property, or of money, so transferred:
- (d) an order releasing, in whole or in part, a security interest given by the company;
- (e) an order requiring security to be given for the discharge of an order made under this clause:
- (f) an order specifying the extent to which a person affected

by the setting aside of a transaction or by an order made under this clause is entitled to claim as a creditor in the liquidation.

10 Additional provisions relating to setting aside transactions and security interests

- (1) The setting aside of a transaction or an order made under clause 9 does not affect the title or interest of a person in property that the person has acquired:
 - (a) from a person other than the company; and
 - (b) for valuable consideration; and
 - (c) without knowledge of the circumstances under which the property was acquired from the company.
- (2) The setting aside of a security interest or an order made under clause 9 does not affect the title or interest of a person in property that the person has acquired:
 - (a) as the result of the exercise of a power of sale by the secured creditor entitled to the security interest; and
 - (b) for valuable consideration; and
 - (c) without knowledge of the circumstances relating to the giving of the security interest.
- (3) Recovery by the liquidator of property or its equivalent value, whether under clause 9 or any other provision of this Act, or under any other enactment, or in equity or otherwise, may be denied wholly or in part if:
 - (a) the person, from whom recovery is sought, received the property in good faith and has altered his or her position in the reasonably held belief that the transfer to that person was validly made and would not be set aside; and
 - (b) in the opinion of the Court, it is inequitable to order recovery or recovery in full.

(4) Nothing in the *Land and Titles Act* restricts the operation of this clause or clauses 4 to 9.

Part 2—Recovery in Other Cases

Division 1—Transactions at Undervalue

11 Definitions

In clause 12:

"specified period" means:

- (a) the period of 1 year before the date of commencement of the liquidation together with the period commencing on that date and ending at the time at which the liquidator is appointed; and
- (b) in the case of a company that was put into liquidation by the Court, the period of 1 year before the making of the application to the Court together with the period commencing on the date of the making of that application and ending on the date on which, and at the time at which, the order of the Court was made: and
- (c) the period of 1 year before the making of the application to the Court together with the period commencing on the date of the making of that application and ending on the date and at the time of the commencement of the liquidation if:
 - (i) an application was made to the Court to put a company into liquidation; and
 - (ii) after the making of the application to the Court a liquidator was appointed under section 16 or 17;

"transaction" includes the giving of a guarantee by a company.

12 Transactions at undervalue

(1) A liquidator of a company may recover from any other party to a transaction any amount by which the value of the

consideration or benefit provided by the company exceeded the value of the consideration or benefit received by the company if:

- (a) the transaction was entered into by a company within the specified period; and
- (b) the value of the consideration or benefit received by the company was less than the value of the consideration provided by the company, or the company received no consideration or benefit; and
- (c) when the transaction was entered into, the company:
 - (i) was unable to pay its due debts; or
 - (ii) was engaged, or about to engage, in business for which its financial resources were unreasonably small; or
 - (iii) incurred an obligation knowing that the company would not be able to perform the obligation when required to do so; and
- (d) when the transaction was entered into, the other party to the transaction knew or ought to have known of the matter referred to in paragraph (c).
- (2) A liquidator of a company may recover from any other party to a transaction any amount by which the value of the consideration or benefit provided by the company exceeded the value of the consideration or benefit received by the company if:
 - (a) the transaction was entered into by a company within the specified period; and
 - (b) the value of the consideration or benefit received by the company was less than the value of the consideration provided by the company, or the company received no consideration or benefit; and

- (c) the company became unable to pay its due debts as a result of the transaction; and
- (d) when the transaction was entered into, the other party to the transaction knew or ought to have known that the company would become unable to pay its due debts as a result of the transaction.

Division 2—Transactions for Inadequate or Excessive Consideration with Directors, etc.

13 Definitions

- (1) In clause 14, "specified period" means:
 - (a) the period of 3 years before the date of commencement of the liquidation together with the period commencing on that date and ending at the time at which the liquidator is appointed; and
 - (b) in the case of a company that was put into liquidation by the Court, the period of 3 years before the making of the application to the Court together with the period commencing on the date of the making of the application and ending on the date on which, and at the time at which, the order of the Court was made; and
 - (c) the period of 3 years before the making of the application to the Court together with the period commencing on the date of the making of that application and ending on the date and at the time of the commencement of the liquidation if:
 - (i) an application was made to the Court to put a company into liquidation; and
 - (ii) after the making of the application to the Court a liquidator was appointed under section 16 or 17; and
- (2) For the purposes of clause 14:
 - (a) the "value of a business or property" includes the value

of any goodwill attaching to the business or property; and

(b) without limiting the circumstances in which a company may be taken to be controlled by a person, a company is controlled by a person, if that person may, by exercising a power exercisable by that person (whether with or without the consent or concurrence of any other person), appoint or remove all the directors of the company, or any number of directors as together hold a majority of voting rights at a meeting of directors.

14 Transactions for excessive consideration with directors, etc.

A liquidator of a company may recover from the person, relative, company, or related company, as the case may be, any amount by which the value of the consideration given for the acquisition of the business, property, or services exceeded the value of the business, property, or services at the time of the acquisition if, within the specified period, the company has acquired a business or property from, or the services of:

- (a) a person who was, at the time of the acquisition, a director of the company, or a nominee or relative of or a trustee for, or a trustee for a relative of, a director of the company; or
- (b) a person, or a relative of a person, who, at the time of the acquisition, had control of the company; or
- (c) another company that was, at the time of the acquisition, controlled by a director of the company, or a nominee or relative of, or a trustee for, or a trustee for a relative of, a director of the company; or
- (d) another company that was, at the time of the acquisition, a related company.

15 Transactions for inadequate consideration with directors, etc.

A liquidator of a company may recover from the person, relative,

company, or related company, as the case may be, any amount by which the value of the business, property, or services, or the value of the shares, at the time of the disposition, provision, or issue exceeded the value of any consideration received by the company if, within the specified period, a company has disposed of a business or property, or provided services, or issued shares, to:

- (a) a person who was, at the time of the disposition, provision, or issue, a director of the company, or a nominee or relative of or a trustee for, or a trustee for a relative of, a director of the company; or
- (b) a person, or a relative of a person, who, at the time of the disposition, provision, or issue, had control of the company; or
- (c) another company that was, at the time of the disposition, provision, or issue, controlled by a director of the company, or a nominee or relative of or a trustee for, or a trustee for a relative of, a director of the company; or
- (d) another company that, at the time of the disposition, provision, or issue, was a related company.

Division 3—Court May Set Aside Certain Securities and Security Interests

16 Court may set aside certain securities and security interests

- (1) The Court may, on the application of the liquidator of a company, order that a security or security interest, or part of it, created by the company over any of its property in favour of any of the persons referred to in subclause (2) must, so far as any security on the property is conferred, be set aside as against the liquidator if:
 - (a) the company is unable to meet all its debts; and
 - (b) the Court considers that, having regard to the circumstances in which the security or security interest was created, the conduct of the person, relative, company, or related company, as the case may be, in

relation to the affairs of the company, and any other relevant circumstances, it is just and equitable to make the order.

- (2) The persons referred to in subclause (1) are as follows:
 - (a) a person who was, at the time the security or security interest was created, a director of the company, or a nominee or relative of, or a trustee for, or a trustee for a relative of, a director of the company; or
 - (b) a person, or a relative of a person, who, at the time when the security or security interest was created, had control of the company; or
 - (c) another company that was, when the security or security interest was created, controlled by a director of the company, or a nominee or relative of, or a trustee for, or a trustee for a relative of, a director of the company; or
 - (d) another company, that at the time when the security or security interest was created, was a related company.

17 Certain securities exempted

Clause 16 does not apply to a security or security interest that has been transferred by the person (person A) in whose favour it was originally created and has been purchased by another person (whether or not from person A) if:

- (a) at the time of the purchase, the purchaser was not a person specified in that clause; and
- (b) the purchase was made in good faith and for valuable consideration.

18 Other orders, etc.

- (1) The Court may make any other orders that it thinks proper for the purpose of giving effect to an order under this clause.
- (2) Nothing in the *Land and Titles Act* restricts the operation of this clause or clauses 16 and 17.

Division 4—Contribution for Not Keeping Proper Accounting Records

19 Contribution for not keeping proper accounting records

- (1) The Court, on the application of the liquidator, may, if it thinks it proper to do so, declare that any 1 or more of the directors and former directors of a company is, or are, personally responsible, without limitation of liability, for all or any part of the debts and other liabilities of the company that the Court may direct if:
 - (a) the company that is in liquidation and is unable to pay all its debts has failed to comply with:
 - (i) section 124 of the *Companies Act 2009* (which relates to the keeping of accounting records); or
 - (ii) section 125 of the *Companies Act 2009* (which relates to the preparation of financial statements); and
 - (b) the Court considers that:
 - the failure to comply has contributed to the company's inability to pay all its debts, or has resulted in substantial uncertainty as to the assets and liabilities of the company, or has substantially impeded the orderly liquidation; or
 - (ii) for any other reason it is proper to make a declaration.
- (2) The Court may give any direction it thinks fit for the purpose of giving effect to the declaration.
- (3) The Court may make a declaration under this clause even though the person concerned is liable to be convicted of an offence.

20 When Court may not make declaration under clause 19

The Court must not make a declaration under clause 19 in relation to a person if the Court considers that the person:

100

- (a) took all reasonable steps to secure compliance by the company with clause 19(1)(a); or
- (b) had reasonable grounds to believe and did believe that a competent and reliable person was charged with the duty of seeing that that provision was complied with and was in a position to discharge that duty.

Division 5—Court May Require Persons to Repay Money or Return Property

21 Court may require persons to repay money or return property

- (1) The Court may, on the application of the liquidator or a creditor or shareholder, do any of the things set out in subclause (2) if, in the course of the liquidation of a company, it appears to the Court that a person who has taken part in the formation or promotion of the company, or a past or present director, manager, liquidator, receiver, or officer of the company, has:
 - (a) misapplied, or retained, or become liable or accountable for, money or property of the company; or
 - (b) been guilty of negligence, default, or breach of duty or trust in relation to the company.

(2) The Court may:

- (a) inquire into the conduct of the promoter, director, manager, liquidator, receiver, or officer; and
- (b) order that person:
 - (i) to repay or restore the money or property or any part of it with interest at a rate the Court thinks just; or
 - (ii) to contribute such sum to the assets of the company by way of compensation as the Court thinks just; or
- (c) if the application is made by a creditor, order that person to pay or transfer the money or property or any part of it, with interest at a rate the Court thinks just, to the creditor.

(3) This clause has effect even though the conduct may constitute an offence.

Division 6—Pooling of Assets

22 Pooling of assets of related companies

- (1) On the application of the liquidator, or a creditor or shareholder, the Court, if satisfied that it is just and equitable to do so, may order that:
 - (a) a company that is, or has been, related to the company in liquidation must pay to the liquidator the whole or part of any or all of the claims made in the liquidation;
 - (b) if 2 or more related companies are in liquidation, the liquidations in respect of each company must proceed together as if they were 1 company to the extent that the Court so orders and subject to such terms and conditions as the Court may impose.
- (2) The Court may make any other order or give any directions to facilitate giving effect to an order under subclause (1) that it thinks fit.

23 Guidelines for orders

- (1) In deciding whether it is just and equitable to make an order under clause 22(1)(a), the Court must consider the following matters:
 - (a) the extent to which the related company took part in the management of the company in liquidation;
 - (b) the conduct of the related company towards the creditors of the company in liquidation;
 - (c) the extent to which the circumstances that gave rise to the liquidation of the company are attributable to the actions of the related company;
 - (d) any other matters as the Court thinks fit.

- (2) In deciding whether it is just and equitable to make an order under clause 22(1)(b), the Court must consider the following matters:
 - (a) the extent to which any of the companies took part in the management of any of the other companies;
 - (b) the conduct of any of the companies towards the creditors of any of the other companies;
 - (c) the extent to which the circumstances that gave rise to the liquidation of any of the companies are attributable to the actions of any of the other companies;
 - (d) the extent to which the businesses of the companies have been combined;
 - (e) any other matters that the Court thinks fit.
- (3) The fact that creditors of a company in liquidation relied on the fact that another company is, or was, related to it is not a ground for making an order under clause 22.

SCHEDULE 8

(Section 13)

CREDITORS' CLAIMS

Part 1—Preliminary Provisions

1 Admissible claims

- (1) Subject to subclause (2), a debt or liability, present or future, certain or contingent, whether it is an ascertained debt or a liability for damages, may be admitted as a claim against a company in liquidation.
- (2) Fines, monetary penalties, and costs to which clause 4 applies are not claims that may be admitted against a company in liquidation.

2 Ascertainment of amount of claim

103

- (1) The amount of a claim must be ascertained as at the date and time of commencement of the liquidation.
- (2) The amount of a claim based on a debt or liability denominated in a currency other than the currency of Solomon Islands must be converted into the currency of Solomon Islands at the rate of exchange on the date of commencement of the liquidation.

3 Claim not of ascertained amount

- (1) If a claim is subject to a contingency, or is for damages, or, if for some other reason the amount of the claim is not certain, the liquidator may:
 - (a) make an estimate of the amount of the claim; or
 - (b) refer the matter to the Court for a decision on the amount of the claim.
- (2) On the application of the liquidator, or of a claimant who is aggrieved by an estimate made by the liquidator, the Court may determine the amount of the claim as it thinks fit.

4 Fines and penalties

Nothing in this Act limits or affects the recovery of:

- (a) a fine imposed on a company, whether before or after the commencement of the liquidation of the company, for the commission of an offence; or
- (b) a monetary penalty payable to the Government imposed on a company by a Court, whether before or after the commencement of the liquidation of the company, for the breach of any written law; or
- (c) costs ordered to be paid by the company in relation to proceedings for the offence or breach.

5 Claims relating to debts payable after commencement of liquidation

(1) A claim in respect of a debt that, but for the liquidation, would

104

not be payable until a date that is 6 months, or later than 6 months, after the date of commencement of the liquidation is to be treated, for the purposes of this Part, as a claim for the present value of the debt.

(2) For the purposes of subclause (1), the present value of a debt is to be determined by deducting from the amount of the debt interest at an appropriate rate for the period from the date on which the company is put into liquidation to the date when the debt is due.

6 Claims by unsecured creditors

- (1) A claim by an unsecured creditor against a company in liquidation must be made in the prescribed form and must:
 - (a) contain full details of the claim; and
 - (b) identify any documents that evidence or substantiate the claim.
- (2) The liquidator may require the production of a document referred to in subclause (1)(b).
- (3) The liquidator:
 - (a) must, as soon as practicable, either admit or reject a claim in whole or in part; and
 - (b) if the liquidator later considers that a claim has been wrongly admitted or rejected in whole or in part, may revoke or amend that decision; and
 - (c) must record in writing any decision made under this subclause.
- (4) If a liquidator rejects a claim, whether in whole or in part, he or she must immediately give notice in writing of the rejection to the creditor.
- (5) The costs of making a claim under subclause (1) or producing a document under subclause (2) must be met by the creditor making the claim.

- (6) A person commits an offence and is liable on conviction to a fine not exceeding 12,500 penalty units or to imprisonment for a term not exceeding 2 years, or both, if the person:
 - (a) makes, or authorises the making of, a claim under this clause that is false or misleading in a material particular knowing it to be false or misleading; or
 - (b) omits, or authorises the omission, from a claim under this clause of any matter knowing that the omission makes the claim false or misleading in a material particular.

Part 2—Secured Claims

7 Powers of secured creditors

- A secured creditor may:
 - (a) realise property subject to a security interest, if entitled to do so; or
 - (b) value the property subject to the security interest and claim in the liquidation as an unsecured creditor for the balance due, if any; or
 - (c) surrender the security interest to the liquidator for the general benefit of creditors and claim in the liquidation as an unsecured creditor for the whole debt.
- (2) A secured creditor may exercise the power referred to in subclause (1)(a) whether or not the secured creditor has exercised the power referred to in subclause (1)(b).

8 Realising secured property

A secured creditor who realises secured property:

- (a) may, unless the liquidator has accepted a valuation and claim by the secured creditor under clause 10, claim as an unsecured creditor for any balance due after deducting the net amount realised;
- (b) must account to the liquidator for any surplus remaining

106

from the net amount realised after satisfaction of the debt, including interest payable in respect of that debt up to the time of its satisfaction, and after making any proper payments to the holder of any other security interest over the property subject to the security interest.

9 Valuation of security

- (1) If a secured creditor values the security and claims as an unsecured creditor for the balance due, if any, the valuation and any claim must be made in the prescribed form and:
 - (a) contain full details of the valuation and any claim; and
 - (b) contain full details of the security interest including the date on which it was given; and
 - (c) identify any documents that substantiate the claim and the security interest.
- (2) The liquidator may require production of any document referred to in subclause (1)(c).

10 Liquidator's duties on receipt of claim by secured creditor

If a claim is made by a secured creditor under clause 9, the liquidator must:

- (a) accept the valuation and claim; or
- (b) reject the valuation and claim, in whole or in part, but:
 - (i) if a valuation and claim is rejected in whole or in part, the creditor may make a revised valuation and claim within 10 working days of receiving notice of the rejection; and
 - (ii) the liquidator may, if he or she later considers that a valuation and claim was wrongly rejected, in whole or in part, revoke or amend that decision; and
- (c) record in writing any decision made by the liquidator under this clause.

11 Liquidator may redeem security

The liquidator may, unless the secured creditor has realised the property, at any time, redeem the security on payment of the assessed value if the liquidator:

- (a) accepts a valuation and claim under clause 10(a); or
- (b) accepts a revised valuation and claim under clause 10(b)(i); or
- (c) accepts a valuation and claim on revoking or amending a decision to reject a claim under clause 10(b)(ii).

12 Liquidator may require secured creditor to exercise powers

- (1) The liquidator may at any time, by notice in writing, require a secured creditor, within 20 working days after receipt of the notice, to:
 - (a) elect which of the powers referred to in clause 7 the creditor wishes to exercise; and
 - (b) if the creditor elects to exercise the power referred to in clause 7(1)(b) or (c), exercise the power within that period.
- (2) A secured creditor on whom notice has been served under subclause (1) who fails to comply with the notice, is to be taken as having surrendered the security interest to the liquidator under clause 7(1)(c) for the general benefit of creditors, and may claim in the liquidation as an unsecured creditor for the whole debt.
- (3) A secured creditor who has surrendered a security interest under clause 7(1)(c) or who is taken as having surrendered a security interest under subclause (2) may, with the leave of the Court or the liquidator and subject to any conditions that the Court or the liquidator thinks fit, at any time before the liquidator has realised the property which is subject to the security interest:

- (a) withdraw the surrender and rely on the security interest;or
- (b) submit a new claim under this clause.

13 Offence to make false or misleading claim

A person commits an offence and is liable on conviction to a fine not exceeding 50,000 penalty units or to imprisonment for a term not exceeding 7 years, or both if the person:

- (a) makes, or authorises the making of, a claim under clause 9 that is false or misleading in a material particular knowing it to be false or misleading; or
- (b) omits, or authorises the omission, from a claim under that clause of any matter knowing that the omission makes the claim false or misleading in a material particular.

Part 3—Preferential Claims

14 Definitions

For the purposes of this Part:

"remuneration" in respect of a period of holiday or of absence from work through sickness or other good cause must be treated as wages in respect of services rendered to the company during that period;

"paid annual leave", in relation to a person, means all sums payable to that person by the company that by or under any written law or any award, agreement, or contract of service are payable to that person by the company as holiday pay.

15 First priority claims

The liquidator must first pay the following, in the order of priority in which they are listed:

 (a) the fees and expenses properly incurred by the liquidator in carrying out the duties and exercising the powers of the liquidator and the remuneration of the liquidator;

- (b) the reasonable costs of a person who applied to the Court for an order that the company be put into liquidation, including the reasonable costs of a person appearing on the application whose costs are allowed by the Court;
- (c) the actual out-of-pocket expenses necessarily incurred by a liquidation committee.

16 Second priority claims

(1) After paying the claims referred to in clause 15, the liquidator must next pay the following claims:

Employees' wages or salary

(a) subject to subclause (2), all wages or salary of any employee, whether or not earned wholly or in part by way of commission, and whether payable for time or for piece work, in respect of services rendered to the company during the 4 months before the commencement of the liquidation;

Employees' annual leave

(b) subject to subclause (2), paid annual leave becoming payable to an employee (or if the employee has died, to any other person in the employee's right) on the termination of the employment before or by reason of the commencement of the liquidation;

Workers compensation

(c) amounts due in respect of any compensation or liability for compensation payable to an employee or to the dependents of an employee that accrued before the commencement of the liquidation;

Amounts deducted by company from employees' wages or salary

(d) subject to subclause (2), amounts deducted by the company from the wages or salary of an employee in order to satisfy obligations of the employee;

Preferential claims under section 35

(e) amounts that are preferential claims under section 35(2);

Superannuation/Provident fund contributions

 (f) any contributions payable by the company to a superannuation scheme or provident fund of an employee;

Priority payments under other enactments

- (g) all sums that by any other enactment are required to be paid in accordance with the priority established by this clause.
- (2) The total sum to which priority is to be given under subclause (1)(a), (b), (d), (e), or (f) must not, in the case of any 1 employee, exceed \$20,000 or any greater amount that is prescribed at the commencement of the liquidation.
- (3) If a payment has been made to the following persons out of money advanced by some person for that purpose, the person by whom the money was advanced has, in a liquidation, the same right of priority in respect of the money advanced as the employee, or other person receiving the payment in right of the employee, would have if the payment had not been made:
 - (a) an employee of a company on account of wages or salary;
 - (b) any such employee or, if the employee has died, to any other person in the employee's right, on account of holiday pay.

17 Third priority claims

After paying the sums referred to in clause 16, the liquidator must next pay the following to the extent that the amount is for the time being unpaid:

Income tax

(a) tax payable by the company;

Tax deductions

 (b) tax deductions made by the company in respect of the earnings of employees of the company;

Customs and excise duty

(c) duty payable within the meaning of the *Customs and Excise Act* (Cap. 121)

18 Ranking of claims in clauses 16 and 17

- (1) The claims listed in each of clauses 16 and 17:
 - (a) rank equally among themselves and must be paid in full, unless the assets are insufficient to meet them, in which case they abate in equal proportions; and
 - (b) so far as the assets of the company available for payment of general creditors are insufficient to meet them, have priority over the claims of persons in respect of assets that are subject to a floating security interest and must be paid accordingly out of those assets.
- (2) To the extent that the claims to which subclause (1) applies are paid out of assets referred to in paragraph (b) of that subclause, the amount so paid is an unsecured debt due by the company to the secured party.

19 When landlord or other person has distrained on goods, etc.

If a landlord or other person has distrained on goods or effects of the company within the month before the commencement of the liquidation:

- the claims to which priority is given by this Part are a first security interest on the goods or effects so distrained on, or the proceeds from their sale; but
- (b) if any money is paid to a claimant under any such security interest, the landlord or other person has the same rights of priority as that claimant.

Part 4—Mutual Credit and Set-off

20 Definitions

In this Part:

"related person" includes a related company and includes a director of the company in liquidation;

"restricted period" means:

- (a) the period of 2 years before the date of commencement of the liquidation together with the period commencing on that date and ending at the time at which the liquidator is appointed; and
- (b) in the case of a company that was put into liquidation by the Court, the period of 2 years before the making of the application to the Court together with the period commencing on the date of the making of that application and ending on the date on which, and at the time at which, the order of the Court was made; and
- (c) the period of 2 years before the making of the application to the Court together with the period commencing on the date of the making of an application and ending on the date and at the time of the commencement of the liquidation if:
 - (i) the application was made to the Court to put a company into liquidation; and
 - (ii) after the making of the application to the Court a liquidator was appointed under section 16 or 17;

"specified period" means:

- (a) the period of 6 months before the date of commencement of the liquidation together with the period commencing on that date and ending at the time at which the liquidator is appointed; and
- (b) in the case of a company that was put into liquidation by the Court, the period of 6 months before the making of the

113

Reprint authorised by the Attorney General and published under the Legislation Act 2023

application to the Court together with the period commencing on the date of the making of that application and ending on the date on which, and at the time at which, the order of the Court was made; and

- (c) the period of 6 months before the making of the application to the Court together with the period commencing on the date of the making of an application and ending on the date and at the time of the commencement of the liquidation if:
 - (i) the application was made to the Court to put a company into liquidation; and
 - (ii) after the making of the application to the Court a liquidator was appointed under section 16 or 17.

21 Mutual credit and set-off

If there have been mutual credits, mutual debts, or other mutual dealings between a company and a person who seeks or, but for the operation of this clause, would seek to have a claim admitted in the liquidation of the company:

- (a) an account must be taken of what is due from the one party to the other in respect of those credits, debts, or dealings; and
- (b) an amount due from one party must be set-off against an amount due from the other party; and
- (c) only the balance of the account may be claimed in the liquidation, or is payable to the company, as the case may be.

22 Proof for person who is not related person

Unless the person proves that, at the time of the transaction or assignment, the person did not have reason to suspect that the company was unable to pay its debts as they became due, a person who is not a related person is not entitled to claim the benefit of a set-off arising from:

- (a) a transaction made within the specified period, being a transaction by which the person gave credit to the company or the company gave credit to the person; or
- (b) the assignment within the specified period to that person of a debt owed by the company to another person.

23 Proof for person who is related person

Unless the related person proves that, at the time of the transaction or assignment, the related person did not have reason to suspect that the company was unable to pay its debts as they became due, the related person is not entitled to claim the benefit of a set-off arising from:

- (a) a transaction made within the restricted period, being a transaction by which the related person gave credit to the company or the company gave credit to the related person; or
- (b) the assignment within the restricted period to that person of a debt owed by the company to another person.

24 Exception for amounts paid or payable by shareholder

Clauses 21 to 23 do not apply to an amount paid or payable by a shareholder or former shareholder:

- (a) as the consideration, or part of the consideration, for the issue of a share; or
- (b) in satisfaction of a call in respect of an outstanding liability of the shareholder made by the directors or by the liquidator.

Part 5—Miscellaneous

25 Interest on claims

- (1) The amount of a claim may include interest up to the date of commencement of the liquidation:
 - (a) at such rate as may be specified or contained in any

115

Reprint authorised by the Attorney General and published under the Legislation Act 2023

- contract that makes provision for the payment of interest on that amount; or
- (b) in the case of a judgment debt, at such rate as is payable on the judgment debt.
- (2) If any surplus assets remain after the payment of all admitted claims, interest must be paid at the prescribed rate on those claims from the date of commencement of the liquidation to the date on which each claim is paid, and if the amount of the surplus assets is insufficient to pay interest in full on all claims, payment must abate rateably among all claims.
- (3) If any surplus assets remain after the payment of interest in accordance with subclause (2), interest must be paid on all admitted claims referred to in subclause (1) from the date of commencement of the liquidation to the date on which the claim is paid at a rate equal to the excess between the prescribed rate and the rate referred to in subclause (1)(a), as the case may be, and, if the amount of the surplus assets is insufficient to pay interest in full on all claims, payment must abate rateably among all claims.
- (4) For the purposes of this clause, prescribed rate means the rate prescribed in regulations made under this Act or, if no such rate is prescribed, the rate at which interest is payable on money due under a judgment of the Court.

26 Trade discounts

A creditor making his or her claim must deduct all trade discounts that he or she would otherwise have given if the company had not gone into liquidation.

27 Periodical payments

(1) When any payment (including rent) falls due at stated periods, and liquidation commences at any time other than at the beginning of 1 of those periods, the persons entitled to the payment may claim up to the date of commencement of liquidation as if the payment accrued on a daily basis. (2) Nothing in subclause (1) affects the right of the lessor of the property to claim rent that accrues on or after the commencement of liquidation.

28 Employees' claims

- (1) A person may make a claim on behalf of all or a number of employees of the company.
- (2) A list setting out the names of the employees, and the amounts due to each of them, must be attached to the claim.
- (3) Any claim made in compliance with this clause has the same effect as if separate claims had been made by each of the employees.

29 Notice to creditors to claim

- (1) Subject to the provisions of this Act, and unless otherwise ordered by the Court, the liquidator may fix a certain day, which must not be less than 10 working days from the date of the notice, on or before which the creditors of the company are to make their claims, and to establish any priority their claims may have under Part 3 of this Schedule.
- (2) The liquidator must give public notice of the day fixed in accordance with subclause (1).

30 Failure to claim by day fixed for claims

- (1) Subject to subclause (2), any creditor who fails to make a claim on or before the day fixed in accordance with clause 29 will be excluded from the benefit of any distribution made before his or her claim is made.
- (2) A creditor who makes a claim after the day fixed in accordance with clause 29 and whose claim is admitted is entitled to receive the benefit of any distribution from which the creditor was previously excluded if any assets remain, or, in the opinion of the liquidator, are likely to remain, available for distribution.

31 Failure to establish priority by day fixed for claims

- (1) Subject to subclause (2), any creditor who fails to establish any priority that the creditor's claim may have on or before the day fixed in accordance with clause 29 must be excluded from objecting to any distribution made before the priority of that claim is established.
- (2) The liquidator may, in making any distribution after the claim is admitted, make an assumption as to the priority that the claim may have and accord the creditor the benefit of the distribution accordingly.
- (3) A creditor who establishes the priority of the creditor's claim after the day fixed in accordance with clause 29 is entitled to receive the benefit of any distribution from which the creditor was previously excluded (if any) if any assets remain, or, in the opinion of the liquidator, are likely to remain, available for distribution.

32 Dividends in respect of rejected claims

- (1) If any creditor applies to the Court for an order reversing or modifying the decision of a liquidator to reject the creditor's claim, the liquidator may in any such case make provision for the dividend on the claim, and the probable cost of the application in the event of the claim being admitted.
- (2) If no notice of an application has been given within the time specified in this Act or in regulations made under this Act for appeals to the Court from a decision of the liquidator, the liquidator must exclude all claims that have been rejected from participation in the dividend.

33 Costs of proceedings relating to liquidator's decision on claim

If any creditor applies to the Court for an order reversing or modifying the decision of a liquidator to reject the creditor's claim, the Court may, if it thinks fit:

(a) allow any costs of any creditor to be added to the creditor's claim;

- (b) allow any costs of any party to be paid out of the assets of the company, such costs being deemed to be expenses of the liquidator;
- (c) order any costs to be paid by any party to the proceedings other than the liquidator.

SCHEDULE 9

(Section 57)

LIQUIDATION OF ASSETS OF OVERSEAS COMPANIES

1 Modified application of Division 2 of Part 2 of this Act

Division 2 of Part 2 of this Act applies to the liquidation of the assets in Solomon Islands of an overseas company, with the following modifications and exclusions:

- references to assets are to be taken as references to assets in Solomon Islands;
- (b) references to a company are to be taken as references to an overseas company;
- (c) references to removal from the Solomon Islands register are to be taken as references to ceasing to carry on business in Solomon Islands;
- (d) the following provisions do not apply to such a liquidation:
 - (i) clause 6 of Schedule 3;
 - (ii) clauses 1 and 3 of Schedule 5;
- (e) clause 2 of Schedule 5 does not affect the tenure of directors of an overseas company, but the overseas company and its directors cease to have any powers, functions, or duties in relation to the company's assets in Solomon Islands, other than those required or permitted to be exercised by Division 2 of Part 2 of this Act;
- (f) section 53 applies to such a liquidation, but instead of

119

Reprint authorised by the Attorney General and published under the Legislation Act 2023

making the statement required by subsection (1)(b)(iii) of that section, the liquidator must state that the company has ceased to carry on business in Solomon Islands and is ready to be removed from the overseas register.

2 Rights of action not affected

Nothing in this Act excludes the right of a creditor of an overseas company in relation to the assets of which a liquidator has been appointed:

- (a) to bring proceedings outside Solomon Islands against the overseas company in relation to a debt not claimed in the liquidation or the balance of a debt remaining unpaid after the completion of a liquidation; or
- (b) to bring an action in Solomon Islands in relation to the balance of a debt remaining unpaid after the completion of a liquidation.

SCHEDULE 10

(Section 59(3) and Clause 3 of Schedule 11)

POWERS, DUTIES, AND LIABILITIES OF RECEIVERS

Part 1—Preliminary Provisions

1 Receiver is grantor's agent

A receiver appointed by, or under a power conferred by, a document is the agent of the grantor unless it is expressly provided otherwise in the document by or under which the receiver was appointed.

2 Exercise of powers by 2 or more receivers

Two or more receivers may act jointly or severally to the extent that they have the same powers unless the document under which, or the order of the Court by which, they are appointed expressly provides otherwise.

Part 2—Powers of Receivers

3 General powers of receivers

120

Subject to the document or the order of the Court appointing the receiver, a receiver may exercise the following powers:

- (a) demand and recover income of the property in receivership;
- (b) issue receipts for income recovered;
- (c) manage the property in receivership;
- (d) insure the property in receivership;
- (e) repair and maintain the property in receivership;
- (f) inspect at any reasonable time documents that relate to the property in receivership and that are in the possession or under the control of the grantor;
- (g) exercise, on behalf of the grantor, a right to inspect documents that relate to the property in receivership and that are in the possession or under the control of a person other than the grantor;
- (h) if the receiver is appointed in respect of all or substantially all of the property of a grantor, change the registered office of the grantor or address at which the documents of the grantor are kept;
- (i) buy and sell the property in receivership in the ordinary course of business.

4 Receiver may execute documents

- (1) A receiver may execute in the name and on behalf of the grantor all documents necessary or incidental to the exercise of the receiver's powers.
- (2) A document signed on behalf of a grantor by a receiver is deemed to have been properly signed by the grantor.

5 Power to claim amounts unpaid on shares

(1) A receiver has the same powers as the directors of a grantor

that is a company or, if the grantor is being wound up or in liquidation, as the directors would have if it was not being wound up or in liquidation, to:

- (a) make claims against the shareholders, or former shareholders, of the company in respect of any uncalled capital that is, or any unpaid amount in respect of any shares issued by the grantor that are, charged under the document by or under which the receiver was appointed; and
- (b) charge interest on, and enforce payment of, any uncalled capital or unpaid amount.
- (2) The exercise of a power under subclause (1) is, as between the shareholders, or former shareholders, of the grantor affected and the grantor, deemed to be a power properly exercised by the directors of the grantor.

6 Receiver may apply for Court order authorising sale of property

- (1) If the consent of a mortgagee is required to the sale of property in receivership and the receiver is unable to obtain that consent, the receiver may apply to the Court for an order authorising the sale of the property, either by itself or together with other assets.
- (2) The Court may make any order that it thinks fit authorising the sale of the property by the receiver if it is satisfied that:
 - (a) the receiver has made reasonable efforts to obtain the mortgagee's consent; and
 - (b) the sale:
 - (i) is in the interests of the grantor and the grantor's creditors; and
 - (ii) will not substantially prejudice the interests of the mortgagee.

(3) An order under this clause may be made on any conditions that the Court thinks fit.

7 Powers of receiver on liquidation

- (1) Subject to subclause (2), a receiver may, unless the Court orders otherwise, be appointed or continue to act as a receiver and exercise all the powers of a receiver in respect of property of a company that has been put into liquidation.
- (2) A receiver holding office in respect of property referred to in subclause (1) may act as the agent of the grantor only:
 - (a) with the approval of the Court; or
 - (b) with the written consent of the liquidator as the case may be.
- (3) A receiver who, by reason of subclause (2), is not able to act as the agent of the grantor does not, by reason only of that fact, become the agent of a person by whom or in whose interests the receiver was appointed.
- (4) A debt or liability incurred by a grantor through the acts of a receiver who is acting as the agent of the grantor in accordance with subclause (2) is not a cost, charge, or expense of the liquidation.

Part 3—Duties of Receivers

Division 1—Duties

8 Principal duties of receivers

- (1) A receiver must exercise the receiver's powers in:
 - (a) good faith and for a proper purpose; and
 - (b) a manner they believe on reasonable grounds to be in the interests of the person in whose interests the receiver was appointed.
- (2) If a receiver appointed under a document acts or does not act

in accordance with any directions given by the person in whose interests the receiver was appointed, the receiver:

- (a) is not in breach of the duty referred to in subclause (1)(b);but
- (b) is still liable for any breach of the duty referred to in subclause (1)(a) or the duty referred to in clause 9.
- (3) Nothing in this clause or clause 9 limits or affects clause 10.

9 Other interests receiver must consider

To the extent consistent with clause 8, a receiver must exercise their powers with reasonable regard to the interests of:

- (a) the grantor; and
- (b) persons claiming, through the grantor, interests in the property in receivership; and
- (c) unsecured creditors of the grantor; and
- (d) guarantors who may be called on to fulfil obligations of the grantor.

10 Duty of receiver selling property

A receiver who exercises a power of sale of property in receivership owes a duty to obtain the best price reasonably obtainable as at the time of sale to:

- (a) the grantor; and
- (b) persons claiming, through the grantor, interests in the property in receivership; and
- (c) unsecured creditors of the grantor; and
- (d) guarantors who may be called on to fulfil obligations of the grantor.

11 Liability for breach of duty imposed by clause 10

Despite any written law or other law or anything contained in the document appointing the receiver:

- (a) a receiver is still liable for a breach of the duty imposed by clause 10 even though the receiver was acting as the grantor's agent or under a power of attorney from the grantor; and
- (b) the receiver is not entitled to compensation or indemnity from the property in receivership or the grantor in respect of any liability incurred by the receiver arising from a breach of the duty imposed by clause 10.

12 Duty in relation to money

A receiver must keep money for the property in receivership separate from:

- (a) other money received in the course of, but not relating to, the receivership; and
- (b) other money held by or under the control of the receiver.

13 Accounting records

- A receiver must, at all times, keep accounting records that correctly record and explain all receipts, expenditure, and other transactions for the property in receivership.
- (2) The accounting records must be retained by the receiver for not less than 7 years after the receivership ends.

14 Duty to notify breaches of Acts

- (1) A receiver of a grantor that is a company who considers that the company or any director or officer of the company has committed an offence against this Act, or the *Companies Act* must report that fact to the Registrar.
- (2) A report made under subclause (1), and any communications between the receiver and Registrar relating to that report, are protected by absolute privilege.

- (3) Nothing in subclause (1) imposes a duty on a receiver to investigate whether any offence of the kind referred to in that subclause has been committed.
- (4) A receiver who fails to comply with subclause (1) commits an offence and is liable on conviction to a fine not exceeding 2,500 penalty units.

Division 2—Failure to Comply with Duties

15 Order to comply

- (1) If the Court is satisfied that there is, or has been, a failure to comply, the Court may:
 - (a) relieve the receiver of the duty to comply, wholly or in part;or
 - (b) without prejudice to any other remedy that may be available in relation to a breach of duty by the receiver, order the receiver to comply to the extent specified in the order.
- (2) No application may be made to the Court in relation to a failure to comply unless:
 - (a) notice of the failure to comply has been served on the receiver not less than 5 working days before the date of the application; and
 - (b) as at the date of the application, there is a continuing failure to comply.

16 Prohibition order

- (1) The Court must make an order prohibiting a person from acting as a receiver for a period not exceeding 5 years if the Court is satisfied that the person is unfit to act as a receiver by reason of:
 - (a) persistent failures to comply; or
 - (b) the seriousness of a failure to comply.

- (2) A person to whom a prohibition order applies must not:
 - (a) act as a receiver in any receivership;
 - (b) act as a liquidator in any liquidation.
- (3) In the absence of special reasons to the contrary, evidence of the following is evidence of persistent failures to comply:
 - that on 2 or more occasions within the preceding 5 years, a court has made an order to comply in respect of the same person; or
 - (b) that on 2 or more occasions within the preceding 5 years, an application for an order to comply has been made in respect of the same person and that in each case the person has complied after the making of the application and before the hearing.

17 Who may apply for orders

An application for an order under clause 15 or 16 or both may be made by:

- (a) the Registrar; or
- (b) a receiver; or
- (c) a person seeking appointment as a receiver; or
- (d) the grantor; or
- (e) a person with an interest in the property in receivership;or
- (f) a creditor of the grantor; or
- (g) a guarantor of an obligation of the grantor; or
- (h) if the grantor is a company, or a liquidator of the grantor; or
- (i) if the grantor is a person who has been adjudged

bankrupt, the Official Receiver of the estate of the grantor.

18 Court orders: general

- (1) In making an order under clause 15 or 16 or both, the Court may:
 - (a) extend the time for compliance; or
 - (b) impose any conditions; or
 - (c) make any ancillary orders.
- (2) A copy of every order made under clause 15 or 16 must, within 10 working days of the order being made, be delivered by the applicant to the Registrar of the Court who must keep it on a public file indexed by reference to the name of the receiver concerned.

Part 4—Liability of Receivers

Division 1—Liability

19 General liability of receiver

- (1) Subject to subclauses (2) and (3), a receiver is personally liable:
 - (a) on a contract entered into by the receiver in the exercise of any of the receiver's powers; and
 - (b) for payment of wages or salary that, during the receivership, accrue under a contract of employment relating to the property in receivership and entered into before the appointment of the receiver if notice of the termination of the contract is not lawfully given within 10 working days after the date of appointment; and
 - (c) for the payment of remuneration under any contract with a director of a grantor if the receiver has expressly confirmed the contract.
- (2) A contract referred to in subclause (1)(a) may exclude or limit the personal liability of a receiver who is not appointed by the

Court.

- (3) The Court may, on the application of a receiver and on any conditions that the Court thinks fit, extend the period within which notice of the termination of a contract is required to be given under subclause (1)(b).
- (4) Every application must be made before the expiry of the 10-day working period.

20 Liability for rent and other payments

- (1) Subject to subclauses (2) and (3), a receiver is personally liable, to the extent specified in subclause (2), for rent and any other payments becoming due under an agreement subsisting at the date of the receiver's appointment relating to the use, possession, or occupation by the grantor of property in receivership.
- (2) The liability of a receiver under subclause (1) is limited to that portion of the rent or other payments which is attributable to the period commencing 10 working days after the date of the appointment of the receiver and ending on whichever is the earlier of:
 - (a) the date on which the receivership ends; or
 - (b) the date on which the grantor ceases to use, possess, or occupy the property.
- (3) The Court may, on the application of a receiver:
 - (a) limit the liability of the receiver to a greater extent than that specified in subclause (2); or
 - (b) excuse the receiver from liability under subclause (1).
- (4) Nothing in subclause (1) or (2):
 - (a) must be taken as giving rise to an adoption by a receiver of an agreement referred to in subclause (1); or
 - (b) renders a receiver liable to perform any other obligation

129

under the agreement.

Division 2—Relief from Liability

21 Receiver is entitled to indemnity

- (1) A receiver is entitled to an indemnity out of the property in receivership in respect of the receiver's personal liability under clause 19 or 20.
- (2) Nothing in this clause or clause 19 or 20:
 - (a) limits any other right of indemnity to which a receiver may be entitled; or
 - (b) limits the liability of a receiver on a contract entered into without authority; or
 - (c) confers on a receiver a right to an indemnity in respect of liability on a contract entered into without authority.

22 Court may relieve receiver from personal liability

- (1) The Court may relieve a person who has acted as a receiver from all or any personal liability incurred in the course of the receivership if it is satisfied that:
 - (a) the liability was incurred solely by reason of a defect in the appointment of the receiver or in the document or order of the Court by or under which the receiver was appointed; and
 - (b) the receiver acted honestly and reasonably; and
 - (c) the receiver ought, in the circumstances, to be excused.
- (2) The Court may exercise its powers under subclause (1) on any conditions that it thinks fit.
- (3) A person in whose interests a receiver was appointed is liable, on any conditions that the Court thinks fit, to the extent to which the receiver is relieved from liability by the Court.

Part 5—Court Supervision of Receivers

23 Court directions

- (1) The Court may, on the application of a receiver:
 - (a) give directions in relation to any matter arising in connection with the carrying out of the receivership by the receiver; or
 - (b) revoke or vary those directions.
- (2) Subject to subclause (3), it is a defence to a claim against a receiver in relation to any act or omission by the receiver that the receiver acted or omitted to act in accordance with a direction given under subclause (1).
- (3) The Court may, on the application of a person referred to in clause 17, order that, by reason of the circumstances in which a direction was obtained under subclause (1), a receiver is not entitled to the protection given by subclause (2).

24 Court orders: receiver's remuneration and validity of appointment

- (1) The Court may, on the application of a person referred to in subclause (2):
 - in respect of any period, review or fix the remuneration of a receiver at a level which is reasonable in the circumstances; or
 - (b) to the extent that an amount retained by a receiver as remuneration is found by the Court to be unreasonable in the circumstances, order the receiver to refund the amount; or
 - (c) declare whether or not a receiver was validly appointed in respect of any property or validly entered into possession or assumed control of any property.
- (2) Any of the following persons may apply to the Court:

- (a) the receiver;
- (b) the grantor;
- (c) a creditor of the grantor;
- (d) a person claiming, through the grantor, an interest in the property in receivership;
- (e) the directors of the grantor or, in the case of a grantor that is in liquidation, the directors of the grantor at the time the liquidator was appointed;
- (f) if the grantor is a company, a liquidator;
- (g) if the grantor is a person who has been adjudged bankrupt, the Official Receiver of the estate of the grantor.
- (3) The Court may, on the application of a person referred to in subclause (2), revoke or vary an order made under subclause (1).

25 Court orders are additional to other Court powers

The powers given by clauses 23 and 24:

- (a) are in addition to any other powers the Court may exercise under this Act or any other Act; and
- (b) may be exercised whether or not the receiver has ceased to act as receiver when the application is made.

SCHEDULE 11

(Section 59(2))

OFFICE OF RECEIVER

1 Persons not to be appointed receiver

The following persons must not be appointed or act as a receiver:

(a) a body corporate;

132

- (b) a person who is under 21 years of age;
- (c) a creditor of the grantor;
- (d) a person who is, or who has within the period of 2 years immediately before the commencement of the receivership been:
 - (i) a director or employee of the grantor; or
 - (ii) a director or employee of a mortgagee of the property in receivership;
- (e) a person who has, or who has had within the period of 2 years before the commencement of the receivership, an interest, whether direct or indirect, in a share issued by the grantor;
- (f) a person who is an undischarged bankrupt;
- (g) a person in respect of whom an order of custody is in force under section 18 or 41 of the *Mental Treatment Act* (Cap 103)
- (h) a person in respect of whom an order has been made under clause 15 or 16 of Schedule 10;
- a person who is prohibited from being a director or promoter of or being concerned or taking part in the management of a company under, the *Companies Act* 2009;
- a person who is disqualified from acting as a receiver by the document that confers the power to appoint a receiver.

2 Validity of acts of receivers

- (1) Subject to subclause (2), no act of a receiver is invalid merely because the receiver was not validly appointed or is disqualified from acting as a receiver or is not authorised to do the act.
- (2) No transaction entered into by a receiver is invalid merely

because the receiver was not validly appointed or is not authorised to enter into the transaction unless the person who deals with the receiver knows, or ought to know, that the receiver was not validly appointed or did not have authority to enter into the transaction.

3 Court may declare whether receiver validly

The Court may, in respect of a person who is disqualified under clause 1 to become or remain a receiver, or fails to comply with an order made under clause 16 of Schedule 10:

- (a) order that the person may be appointed and act or may continue to act as a receiver, despite clause 1; or
- (b) remove the person from office.

4 Court orders for protecting property on

The Court may, on making an order that removes, or has the effect of removing, a receiver from office, make any orders that the Court thinks fit:

- (a) for preserving the property in receivership;
- (b) requiring the receiver for that purpose to make available to any person specified in the order any information and documents in the possession or under the control of the receiver.

5 Vacancy in office of receiver

The office of receiver becomes vacant if the person holding office resigns, dies, or is or becomes disqualified under clause 1.

6 How receiver may resign

(1) A receiver appointed under a power conferred by a document may resign office by giving not less than 5 working days' written notice of the receiver's intention to resign to the person who appointed the receiver. (2) A receiver appointed by the Court must not resign office without first obtaining the leave of the Court to do so.

7 Notice of vacancy

- (1) A person who vacates office as receiver (or if that person is unable to act, that person's legal representative) must:
 - (a) if the vacancy arose because of the disqualification of the person holding office as receiver, immediately notify the person who appointed the receiver; and
 - (b) if the receiver held office in relation to the property of a company, within 5 working days of the vacancy occurring, give written notice of the vacancy to the Registrar for registration in the register of security interests kept by the Registrar; and
 - (c) in all cases, immediately give public notice of the vacancy.
- (2) A person who fails to comply with subclause (1) commits an offence and is liable on conviction to a fine not exceeding 1,250 penalty units.

8 Vacating receiver's successor to be helped

- (1) A person vacating the office of receiver must, if practicable, provide any information and give any assistance in the conduct of the receivership to the receiver's successor as that person reasonably requires.
- (2) On the application of a person appointed to fill a vacancy in the office of receiver, the Court may make any order that the Court considers necessary or desirable to facilitate the performance of the receiver's duties.

ENDNOTES

1 KEY

amd = amended Pt = Part

Ch = Chapter rem = remainder
Div = Division renum = renumbered

exp = expires/expired rep = repealed
GN = Gazette Notice Sch = Schedule
hdg = heading Sdiv = Subdivision

ins = inserted SIG = Solomon Islands Gazette

It = long title st = short title
LN = Legal Notice sub = substituted

nc = not commenced

2 LIST OF LEGISLATION

Companies (Insolvency and Receivership) Act 2009 (No. 2 of 2009)

Assent date 8 May 2009 Gazetted 9 April 2009 Commenced 1 July 2010

Penalties Miscellaneous Amendments Act 2009 (No. 14 of 2009)*

Assent date 29 July 2009
Gazetted 1 October 2009
Commenced 1 October 2009

*[Note: The formula in s5 of the Penalties Miscellaneous Amendments Act 2009 is invalid due to an error which has been corrected in the Legislation Amendment Repeal and Validation Act 2023, and therefore with effect from the commencement of that

Act - 5 Feb 2024]

Companies Act 2009 (No. 1 of 2009)

Assent date 2 April 2009 Gazetted 9 April 2009 Commenced 1 July 2010*

[*Note: s209(2) provides that any reference to the term 'charge' shall be changed to 'security interest' on the commencement of the Secured Transactions Act 2008. That Act commenced on 31

August 2009].

Legislation Amendment, Repeal and Validation Act 2023 (No. 17 of 2023)

Assent date 29 December 2023 Gazetted 29 December 2023 Commenced 5 February 2024

3 LIST OF AMENDMENTS

```
amd by Act No. 1 of 2009
s 10
                amd by Act No. 1 of 2009
s 13
s 15
                amd by Act No. 17 of 2023
                amd by Act No. 1 of 2009
s 20
                amd by Act No. 1 of 2009
s 22
s 27
                amd by Act No. 17 of 2023
                amd by Act No. 1 of 2009
s 36
                amd by Act No. 1 of 2009
s 50
                amd by Act No. 17 of 2023
s 56
s 61
                amd by Act No. 17 of 2023
s 64
                amd by Act No. 17 of 2023
s 65
                amd by Act No. 17 of 2023
s 68
                amd by Act No. 17 of 2023
s 70
                amd by Act No. 17 of 2023
s 74
                amd by Act No. 17 of 2023
s 77 hdg
                amd by Act No. 1 of 2009
                amd by Act No. 1 of 2009
s 77
                amd by Act No. 17 of 2023
s 78
s 79
                amd by Act No. 1 of 2009
s 80
                rep by Act No. 14 of 2009
Sch 2
                amd by Act No. 1 of 2009
Sch 3
                amd by Act No. 1 of 2009
Sch 5
                amd by Acts No. 1 of 2009; No. 17 of 2023
Sch 7
                amd by Act No. 1 of 2009
Sch 8
                amd by Acts No. 1 of 2009; No. 17 of 2023
Sch 10
                amd by Act No. 17 of 2023
Sch 11
                amd by Acts No. 1 of 2009; No. 17 of 2023
```